

# TAX AVOIDANCE, FINANCIAL EXPERTS ON THE AUDIT COMMITTEE, AND BUSINESS STRATEGY

#### Pei-Hui Hsu

College of Business

California State University, East Bay

Hayward, CA 94542-3067

pei-hui.hsu@csueastbay.edu

Jared A. Moore\*

College of Business

Oregon State University

443 Austin Hall, Corvallis, OR 97331-2603

jared.moore@bus.oregonstate.edu

# **Donald O. Neubaum**

College of Business

Florida Atlantic University

777 Glades Road, Boca Raton, FL 33431

dneubaum@fau.edu

This version: August 2018

This article has been accepted for publication and undergone full peer review but has not been through the copyediting, typesetting, pagination and proofreading process, which may lead to differences between this version and the <u>Version of Record</u>. Please cite this article as <u>doi:</u> 10.1111/jbfa.12352.

We appreciate the helpful comments of Andrew Stark (joint-editor) and an anonymous reviewer. Jared Moore gratefully acknowledges financial support from the Mary Ellen Phillips professorship.

#### **Abstract**

We examine whether financial expert audit committee members tailor their approach to overseeing the corporate tax planning process according to the firm's business strategy. We predict and find that such directors encourage defender-type firms (characterized partially by high risk aversion) to engage in more tax avoidance activities and prospector-type firms (characterized partially by innovation and risk seeking) to scale back on tax avoidance, relative to the opposing strategy type. We also find that both accounting experts and non-accounting financial experts on the audit committee contribute to our results to some extent, although the effects of non-accounting financial experts present more consistently. Overall, our results suggest that financial experts on the audit committee tend to play more of an advising role for defenders and more of a monitoring role for prospectors, relative to one another.

**Keywords:** Tax avoidance, tax aggressiveness, financial experts, board of directors, audit committee, business strategy

JEL Classifications: H25, H26, M41, G3, L19, L21

#### 1. INTRODUCTION

This study investigates whether the advising and monitoring roles of the board of directors (Adams and Ferreira 2007) vary with a firm's business strategy in the context of tax avoidance. Specifically, we examine how financial expertise on the audit committee and the firm's strategy type (Higgins et al. 2015; Miles et al. 1978) interact to influence the firm's tax planning activities. Audit committees have come to play an increasingly vital role in firms' tax planning and tax risk management processes (Deloitte 2014; Deloitte 2013; Richardson

et al. 2013; Deloitte 2011). The importance of financial expertise on the audit committee has also increased as the scope of the audit committee's oversight has evolved to include areas such as tax issues, information technology, and globalization, each carrying their own complexities and risks (KPMG 2017; KPMG 2015; KPMG 2012).

In a prior study, Robinson et al. (2012) examine the impact on tax avoidance of financial expertise on the audit committee generally, finding a positive association between financial expertise and *levels* of tax avoidance (consistent with directors' advising role) but a negative association between financial expertise and the use of *risky* tax planning strategies (consistent with directors' monitoring role). What is not revealed from these findings, however, is the extent to which board, specifically audit committee, members vary their application of these roles based on contextual factors or conditions, such as the strategic profile of the firm. We explore this question, thereby helping to provide a richer understanding of how board members approach oversight of the corporate tax planning process.

Prior studies argue that there is an optimal level of tax avoidance for each firm that maximizes firm value (Armstrong et al. 2015; Chen et al. 2010; Slemrod 2004). Recent empirical evidence by Cook et al. (2017) supports this argument; they find that firms' ex ante cost of capital increases with the degree to which their level of tax avoidance deviates from investor expectations in either direction. From an agency theory perspective, in the absence of sufficiently strong incentive alignment and/or monitoring, managers may choose to engage in more or less tax avoidance than would maximize shareholder value (e.g., Rego and Wilson 2012; Desai and Dharmapala 2006). While Fama and Jensen (1983) emphasize

the general importance of the board of directors as a monitoring body, Adams and Ferreira (2007) characterize the board as serving *dual* roles as both monitors of and advisors to management, providing expert council to managers for optimal decision-making while also constraining managers' actions to the extent that they do not benefit shareholders.

The mixed extant evidence linking tax avoidance with board characteristics is indicative of the presence of both roles in the corporate tax planning process. For example, in the context of noncompliant tax avoidance, Lanis and Richardson (2011) and Richardson et al. (2013) find that a higher proportion of independent directors on the board is negatively associated with tax aggressiveness, consistent with the monitoring role. However, Minnick and Noga (2010) find no relation between board independence and either GAAP or cash effective tax rates (ETRs) but a negative association between director pay-performance-sensitivity and GAAP ETRs, more reflective of the advising role at some level. More recently, Armstrong et al. (2015) find that more independent and more sophisticated boards appear to encourage more (less) aggressive tax positions for firms at the lower (upper) extreme of the tax avoidance continuum, consistent with both roles.

As noted previously, Robinson et al. (2012) also find evidence highlighting both the advising and monitoring roles of board members in influencing firms' tax avoidance, specifically those on the audit committee with accounting-specific financial expertise. Their evidence that financial expertise on the audit committee is important to the tax planning process is consistent with the increased role of the audit committee in managing tax risk

<sup>&</sup>lt;sup>1</sup> Richardson et al. (2013) also find the same association using the proportion of independent directors on the audit committee.

(e.g., Deloitte 2014; Deloitte 2013; Richardson et al. 2013; Deloitte 2011) and the growing importance of financial expertise on the part of audit committee members due to risks associated with emerging areas of audit committee oversight (e.g., tax issues, information technology, and globalization). Given the complexity surrounding tax issues, audit committee members with financial expertise are indeed in a unique position to monitor and advise executives' decisions concerning their firms' tax planning activities and the related risks. Similar to Robinson et al. (2012), our analyses explore how audit committee members, specifically financial experts, fulfill their dual advisory and monitoring roles with respect to tax planning.<sup>2</sup> However, our study differs from theirs in that we investigate how a firm's business strategy conditions financial expert audit committee members to tailor how they deliver on these roles.

Miles et al. (1978) develop a typology of four business strategy types, namely defenders, analyzers, prospectors, and reactors, each with different approaches to competing in the marketplace and propensities toward innovation and risk-taking. Higgins et al. (2015) examine the association between tax avoidance and firm strategy type, finding that (more risk seeking) prospectors engage in more tax avoidance and do so in more risky and uncertain ways than (more risk averse) defenders. These findings suggest (as expected) that the different characteristics of each strategy type, such as strategic focus, risk tolerance, and

<sup>&</sup>lt;sup>2</sup> We follow the final definition adopted by the SEC, under which an audit committee member can be deemed a financial expert if the member has: (a) accounting expertise from work experience as a certified public accountant, auditor, CFO, financial comptroller, financial controller, or accounting officer; (b) finance expertise from work experience as an investment banker, financial analyst, or any other financial management role; or (c) supervisory expertise from supervising the preparation of financial statements (e.g., CEO or company president).

organizational structure, influence the manner in which managers weigh the costs and benefits of tax planning.

The board of directors, of which the audit committee is a subset, plays a key role in establishing and maintaining the business strategy of the firm and the general premise of managerial decision making (Baysinger and Hoskisson 1990; Mizruchi 1983). The board accomplishes this partially through the implementation of decision control systems chosen based on the business-level strategy of the firm (Baysinger and Hoskisson 1990) and through domain-specific advising, monitoring, and evaluation of managers' decisions at the committee level (Zahra and Pearce 1989; Jemison and Oakley 1983). Within the committee structure, board members integrate their own skills and expertise with their understanding of the operations and strategies of the firm and apply their expertise in evaluating executives or making recommendations about firm-specific issues within the domain of the committee (Forbes and Milliken 1999; Zahra and Pearce 1989). Thus, we expect that financial expert audit committee members integrate their skills and expertise with their awareness of the firm's business strategy in performing their monitoring and advising roles on issues subject to audit committee oversight, including tax planning. Accordingly, our general hypothesis is that financial experts on the audit committee are likely to act in more of an advising capacity in the tax planning process for defenders and in more of a monitoring capacity in the tax planning process for prospectors.

Following Higgins et al. (2015), we partition our sample into defender, analyzer, and prospector strategy types based on the business strategy score from Bentley et al. (2013), focusing primarily on defenders and prospectors (the opposite ends of the strategy

continuum). We then examine whether the association between the proportion of independent financial experts on the audit committee and tax avoidance differs across strategy types, using multiple measures of tax avoidance to capture both tax planning broadly and the use of more risky tax avoidance tactics. We find that financial expertise on the audit committee is positively associated with tax avoidance activity for defender-type firms in some cases (e.g., cash effective tax rates) and that the association between financial expertise on the audit committee and tax avoidance is consistently significantly more (and generally *net*) negative for prospector-type firms relative to defender-type firms.

Additionally, we differentiate between accounting and non-accounting financial expertise and find that both play a role in the tax planning process. Finally, while the effects of non-accounting financial experts appear more prevalently across our tax avoidance measures, they do not dominate those of accounting financial experts when compared directly.

Overall, our results suggest that financial experts on the audit committee generally play more of an advising role (i.e., more encouraging of tax avoidance) for defender-type firms and more of a monitoring role (i.e., more discouraging of tax avoidance) for prospector-type firms, relative to each other, consistent with our hypothesis. These findings imply not only that financial experts on the audit committee play both advising and monitoring roles in a firm's tax planning process (Robinson et al. 2012), but also that they apply these roles *in the context of the firm's business strategy*. Further, the results suggest that both accounting

<sup>&</sup>lt;sup>3</sup> We use cash effective tax rates to represent tax avoidance broadly and unrecognized tax benefits to represent risky tax planning. We also include total and permanent book-tax differences because they contain information about both the broader impact of tax avoidance on income tax expense and more aggressive tax avoidance.

and non-accounting financial expert audit committee members influence tax planning to some extent, consistent with prior studies which document that both types of financial expertise benefit the financial reporting process (Zhang et al. 2007; Dhaliwal et al. 2010).

Our study contributes to the literature in several ways. We add to the literature investigating the determinants of tax avoidance in an agency theory context (Hanlon and Heitzman 2010), particularly the branch focusing on corporate governance mechanisms. Prior studies link tax avoidance with board characteristics (e.g., Armstrong et al. 2015; Richardson et al. 2013; Robinson et al. 2012; Lanis and Richardson 2011; Minnick and Noga 2010), collectively (and in some cases, individually) highlighting the dual roles of board members as advisors to and monitors of management. Our study builds on these findings as we show that financial expert audit committee members alter the manner in which they perform their advisory and monitoring roles with respect to tax planning according to the firm's business strategy.

We also contribute to the organizational theory literature examining firm strategy. The extant evidence suggests that the different characteristics of firms of the various strategy types influence firms' operational approaches and tolerance for risk and uncertainty, thus impacting operational decisions, including tax avoidance (Higgins et al. 2015). We extend Higgins et al. (2015) by documenting that certain corporate governance mechanisms influence managerial decision making differentially across business strategy types.

Finally, we add to the literature examining the impact of board attributes, including financial expertise, on firms' operational and reporting outcomes. Results from previous studies indicate that board characteristics influence performance around corporate

activities associated with agency issues, such as mergers and acquisitions, earnings management, and tax avoidance (e.g., Byrd and Hickman 1992; Xie et al. 2003; Klein 2002; Minnick and Noga 2010; Lanis and Richardson 2011). Prior evidence further suggests that financial expert board members play a significant monitoring role over and above that of non-financial expert board members (e.g., Robinson et al. 2012; Dhaliwal et al. 2010; Krishnan and Visvanathan 2008; Zhang et al. 2007; Farber 2005). Our results imply that financial expert board (i.e., audit committee) members vary the manner in which they carry out their oversight duties based on high-level firm traits, such as business strategy. An implication of our findings is that future studies examining the impact of board characteristics on firm behavior may benefit from considering the influence of business strategy in their analyses.

Section 2 provides a review of the prior literature and the development of our hypothesis. Section 3 discusses our research methods and data used to test how business strategy impacts the relation between financial expertise on the audit committee and tax avoidance. Section 4 presents the results of our analyses, and Section 5 presents our concluding remarks.

#### 2. BACKGROUND AND HYPOTHESIS DEVELOPMENT

#### 2.1 Tax Avoidance and the Audit Committee

Slemrod (2004) argues that an optimal level of tax planning/avoidance exists for every firm that balances resulting benefits and costs in such a way as to maximize shareholder value. Cook et al. (2017) provide empirical evidence along this line by showing a positive association between the deviation of a firm's tax avoidance level from investor expectations (in either direction) and the firm's ex ante cost of capital. Relatively little is known about what drives a firm's optimal level of tax avoidance or why firms differ in how aggressively they approach tax planning and reporting (Hanlon and Heitzman 2010; Slemrod 2004). While much of the existing literature focuses firm-level characteristics associated with tax avoidance, several recent studies examine tax avoidance/aggressiveness in an agency context using data from various countries around the world, with some exploring the role of various corporate governance dimensions as determinants. Recent work, for example, links tax avoidance (directly or indirectly) with executive compensation (e.g., Rego and Wilson 2012; Minnick and Noga 2010; Desai and Dharmapala 2006; Phillips 2003), institutional ownership (e.g., Khurrana and Moser 2013; Chen et al. 2010; Desai and Dharmapala 2006), family firm status (e.g., Moore et al. 2017; Steijvers and Niskanen 2014; Chen et al. 2010), and governance-related corporate social responsibility traits (e.g., Huseynov and Klamm 2012).

A component of corporate governance that has received less attention in the tax avoidance literature is the board of directors. From an agency theory perspective, boards of

directors play a critical role by establishing and maintaining the business strategy of the firm, selecting and compensating executives, and monitoring executives' decisions to ensure shareholder maximization (Baysinger and Hoskisson 1990; Fama and Jensen 1983).

Empirical studies to date have focused on board monitoring as a means of mitigating agency issues. For example, several papers link board characteristics with performance around specific corporate activities in which agency conflicts are most likely to occur, such as mergers and acquisitions (Byrd and Hickman 1992), CEO replacement (Weisbach 1988), takeover bids (Cotter et al. 1997), and earnings management (e.g., Xie et al. 2003; Klein 2002). Adams and Ferreira (2007) further expand our view of boards of directors by highlighting that they serve *dual* roles as both monitors of and advisors to management, providing expert consultation to help push managers toward optimal outcomes while also constraining managers' actions to the extent that they do not benefit shareholders.

Tax planning provides an interesting setting in which to study boards of directors in the context of Slemrod (2004) and Cook et al. (2017) since, left to their own devices, managers may opt to engage in more or less tax avoidance than would otherwise be optimal from the perspective of shareholders. On the one hand, risk-averse managers may avoid tax avoidance activities that involve significant uncertainty, even if the activities are expected to generate net benefits for shareholders (Rego and Wilson 2012). On the other hand, managers may engage in complex and risky tax avoidance activities, possibly as a means of facilitating and/or hiding extraction of private benefits (Desai and Dharmapala 2006).

Accordingly, to the extent that the board of directors plays a role in tax planning/avoidance, it is an empirical question whether the monitoring or advising role dominates on average.

The extant empirical evidence on this question is mixed. In the context of noncompliant tax avoidance among Australian firms, Lanis and Richardson (2011) and Richardson et al. (2013) find that a higher proportion of independent directors on the board is negatively associated with tax aggressiveness, consistent with the monitoring role. Similarly, using a sample of Finnish firms, Steijvers and Niskanen (2014) find that the presence of outside directors on the board limits tax aggressiveness on the part of non-family CEOs in private family firms. On the other hand, Moore et al. (2017) find a positive association between board independence and tax avoidance using a sample of U.S. firms, while Chan et al. (2013) and Minnick and Noga (2010) find no association at all in Chinese and U.S. settings, respectively. Chan et al. (2013) provide additional evidence consistent with the advising role with their finding of a positive relation between tax avoidance and the percentage of shares held by board members. Minnick and Noga (2010) also provide some evidence of the advising role by documenting a negative relation between director pay-performancesensitivity and GAAP ETRs. Armstrong et al. (2015) find evidence consistent with both roles; more independent and more sophisticated boards appear to encourage more (less) aggressive tax positions for firms at the lower (upper) extreme of the tax avoidance continuum.

We examine how directors, specifically independent financial experts, on the audit committee fulfill their monitoring and advising roles with respect to a firm's tax avoidance activities. Regulators emphasize the importance of financial expertise on the audit committee in the context of financial reporting (e.g., Blue Ribbon Committee on Improving

<sup>4</sup> As noted in footnote 2, we follow the final definition of "financial expert" adopted by the SEC.

the Effectiveness of Corporate Audit Committees 1999; Public Oversight Board 1993; Sarbanes-Oxley Act of 2002 (SOX) 2002),<sup>5</sup> and several recent studies associate financial expertise on the audit committee with financial reporting outcomes such as accruals quality, conservatism, and SEC enforcement actions (e.g., Dhaliwal et al. 2010; Krishnan and Visvanathan 2008; Zhang et al. 2007; Farber 2005). However, the purview of the audit committee has expanded in recent years to include areas such as risk management, information technology, and globalization, and the audit committee is also now viewed as playing an important role in the management of tax risk (Deloitte 2014; Deloitte 2013; Richardson et al. 2013; Deloitte 2011). Moreover, as the scope of the audit committee has expanded to include these areas, the risks associated with them have increased the value and importance of financial expertise possessed by audit committee members (KPMG 2017; KPMG 2015; KPMG 2012). Accordingly, and particularly in light of the complexity surrounding tax issues, audit committee members with financial expertise are in a unique position to oversee and evaluate a firm's tax planning activities, the related risks, and (as discussed below) the appropriateness of its tax planning given the firm's business strategy. In a previous study, Robinson et al. (2012) examine the impact of financial expertise on

the audit committee on tax avoidance in terms of both overall levels of tax avoidance and

<sup>&</sup>lt;sup>5</sup> Section 407 of SOX requires firms to disclose whether at least one of the audit committee members is a financial expert. The NYSE requires that at least one member of the audit committee have accounting or related financial management expertise and that all members of the audit committee be financially literate. The NASDAQ rules require companies to certify whether at least one member of the audit committee has past employment experience in finance or accounting, requisite professional certification in accounting, or any other comparable experience or background that results in the individual's financial sophistication, including being or having been a CEO, CFO, or other senior officer with financial oversight responsibilities.

the use of risky/uncertain tax avoidance strategies. They find a positive association between levels of accounting-specific financial expertise on the audit committee and levels of tax avoidance but a negative association between levels of accounting-specific financial expertise on the audit committee and the propensity toward employing risky tax planning strategies. These results suggest that financial experts on the audit committee serve an advising role with respect to tax planning overall but a monitoring role with respect to the degree of risk taken on by the firm in the tax planning process. We, however, argue that audit committee members vary how they perform these roles according to the firm's business-level strategy, which reflects its tolerance for risk and how it weighs costs and benefits in decision making processes. Thus, we extend the prior literature by investigating the advising and monitoring roles of financial experts on the audit committee with respect to tax planning in the context of the firm's business strategy.

# 2.2 Firm Business Strategy

The term business strategy describes how firms choose to compete and the strategic types that arise from the way firms decide to resolve three fundamental managerial problems: the entrepreneurial problem, the engineering problem, and the administrative problem (Miles et al. 1978). The entrepreneurial problem relates to how a firm should define the goods or services it will provide and the target markets or competitive domains in which it will compete. The engineering problem is related to the creation and implementation of the operational solution to address the entrepreneurial problem. The administrative problem considers how a company should rationalize and stabilize the organizational activities created to solve both the entrepreneurial and engineering problems.

Based on their research, Miles et al. (1978) develop a typology of four strategic types of organizations, namely defenders, analyzers, prospectors, and reactors, all of which attempt to simultaneously solve these three managerial problems.

Defender strategic types tend to compete by offering a limited set of products and/or services to narrow subsets or customer niches within the larger market. These firms seek market dominance through incremental growth of market share in their existing markets, often through competitive pricing tactics. By targeting stable and defensible market domains, defenders solve the engineering problem by focusing their efforts on producing and distributing goods and services in the most efficient and cost-effective manner possible. This strategic emphasis on increasing operational efficiency and stability often leads to a culture of careful planning, risk aversion, and uncertainty avoidance. On the other hand, prospectors, which lie at the opposite end of the strategic continuum, tend to compete by continually finding and exploiting new product and market opportunities. These firms compete broadly and seek market dominance through innovation and product and market development. For these firms, developing and maintaining a reputation as an innovator may be even more important than maintaining high levels of profitability. Prospectors take risks and are flexible, adaptive organizations, which enable them to quickly respond to new opportunities and changing competitive landscapes. Relative to defenders, prospectors are less concerned with minimizing costs and are more focused on experimentation and innovation. Analyzers, the third strategic type identified by Miles et al (1978), possess characteristics shared by both defenders and prospectors and attempt to maintain market share while also being innovative. Finally, reactors do not follow a consistent or deliberate

strategy and are considered a dysfunctional organizational form (Zahra & Pearce 1990).

Thus, reactor firms are almost always excluded in empirical studies using the Miles and

Snow typology (including the current study).

Consistent with these strategic types, we would expect managers of defenders to be more conservative than others and more sensitive to the uncertainty associated with tax avoidance, the potential for tax authorities to levy penalties, and the potential damage to their reputation associated with public disclosure of involvement in particularly egregious avoidance activities. As a result, left to their own devices, managers following a defender strategy are more likely to forgo tax avoidance opportunities viewed as costly or risky, even if the tax avoidance would likely ultimately benefit shareholders. Conversely, the strategic emphasis for prospectors often leads to a culture of competitive aggressiveness and risk taking, which is likely to permeate into these firms' tax avoidance activities. In addition, prospectors are less likely to formally plan, a factor which may cause them to put less weight on careful analysis of the costs and benefits of tax planning as they opt to engage in more risky tax avoidance strategies. Managers of prospectors are likely to be less sensitive to the uncertainties associated with aggressive tax avoidance and more likely to have a higher tolerance threshold for concerns associated with potential tax penalties or threats to the firm's reputation due to public disclosures of serious tax violations. Left to their own devices, managers of firms following a prospector strategy are therefore more likely to be aggressive with their tax planning behavior.<sup>6</sup>

-

<sup>&</sup>lt;sup>6</sup> The three strategic types considered in this study represent a strategy continuum with defenders on one end, prospectors on the other, and analyzers in the middle. Our analyses focus specifically

Higgins et al. (2015) explore the association between tax avoidance and firm business strategy using the Miles et al. (1978) framework. Consistent with these characterizations of the defender and prospector strategy types, they find that prospectors engage in more tax avoidance and do so in more risky and uncertain ways than defenders. They attribute these results to differences in how firms weigh the costs, benefits, and risks of tax planning across different business strategy types.

## 2.3 Financial Expertise on the Audit Committee and Firm Business Strategy

As discussed above, we extend this work and argue that financial experts on the audit committee fulfill their monitoring and advising roles in the tax planning process in the context of the firm's business strategy. This is because the board of directors, of which the audit committee is a subset, plays a key role in establishing and maintaining the strategic direction of the firm and the general premise of managerial decision making (Baysinger and Hoskisson 1990; Mizruchi 1983). Fama and Jensen (1983) discuss the importance of boards' ratification and monitoring of managers' decisions. However, boards' influence on firm outcomes can result from a variety of activities, including establishing goals and objectives to direct executives' attention to key performance outcomes consistent with shareholders' expectations, as well as through strategic decision making (Zahra and Pearce 1989; Mizruchi 1983). As such, boards perform not only a control role (e.g., protection of shareholders' interests), but also provide service (e.g., advice and counsel) and strategic (e.g., involvement

on defenders and prospectors because analyzers possess characteristics of both of the other strategy types (Higgins et al. 2015). Thus, we make no prediction about how this strategy type will interact with financial expertise on the audit committee as a determinant of tax avoidance.

in and articulation of mission and strategy, effective control of strategy) support to executives (Carpenter 1988; Baysinger and Butler 1985; Pfeffer and Salancik 1978).

One mechanism through which the board maintains the strategic direction of the firm and the general premise of managerial decision making is the decision control system employed (Baysinger and Hoskisson 1990). Broadly speaking, boards of directors have two types of decision control systems at their disposal, namely strategic controls and financial controls (Hoskisson and Hitt 1988; Gupta 1987). In financial control systems, managers are almost exclusively evaluated objectively in terms of the firm's ex post financial performance. In strategic control systems, managers are largely evaluated subjectively in terms of the quality of their decisions ex ante, but are also evaluated on their ex post financial objectives. The decision control system a board of directors relies upon while performing its oversight and premise setting roles is a function of the firm's business level strategy, as the business strategy informs the board's approach to evaluating executives and the quality of the decisions executives make.<sup>7</sup>

Because executives are aware of the decision control system utilized by the board of directors, their risk taking postures, as well as their effort and the direction of that effort,

<sup>&</sup>lt;sup>7</sup> For example, when a firm pursues a high risk strategy which requires significant investment in uncertain R&D activities (such as those generally pursued by prospector firms), the board of directors recognizes that managers are merely "influencers of events rather than the controllers of certain outcomes" (Bourgeois 1987, p. 347). As such, board members tend to rely upon strategic controls when they evaluate whether or not the decisions and actions of executives were appropriate within the context of the firm's propensity toward risk taking, or if the actions were the result of incompetence or opportunism. Similarly, when a firm pursues a less aggressive, more predictable and stable strategy (such as those generally pursued by defenders), the board's evaluation of executives relies more on financial controls, judging executives based on their achievement of financial objectives.

are affected (Hambrick and Snow 1989; Eisenhardt 1985). As the board of directors increases its reliance on financial controls, executives increasingly focus on short-term financial performance, and their efforts are potentially directed away from optimally risky strategies (e.g., more aggressive tax positions) which may benefit shareholders. As the board of directors increases its reliance on strategic controls, executives are inclined to pursue more (potentially overly) risky actions and decisions (e.g., more aggressive tax positions), and executives assign higher utility to uncertain cash flows and outcomes.

It is primarily within the committee structure that the appropriateness of managers' domain-specific decisions are evaluated and where domain-specific advising and monitoring roles are performed (Zahra and Pearce 1989; Jemison and Oakley 1983). When board committees are effective in fulfilling these roles, board members integrate their own skills and expertise (e.g., financial expertise) with their awareness of the operations and strategies of the firm and apply their expertise when they evaluate executives or make recommendations about firm-specific issues or concerns within the domain of the committee (Forbes and Milliken 1999; Zahra and Pearce 1989). Accordingly, given the complexity surrounding tax issues, we argue that audit committee members with financial expertise are in a position to oversee and evaluate the appropriateness of a firm's tax planning activities and the related risks given the firm's business strategy.

In the case of defenders, which are naturally risk averse and thus may tend to underinvest in tax avoidance from a value maximization standpoint (Higgins et al. 2015), we expect financial experts on the audit committee to take on more of an advising role, encouraging managers to engage in more tax avoidance activity than they would if left to

their own devices. Prospectors, on the other hand, are naturally prone to innovation and risk taking behaviors and thus may engage in more aggressive or uncertain tax avoidance than is optimal from a shareholder perspective (Higgins et al. 2015). Accordingly, we expect financial experts on the audit committee of a prospector firm to take on more of a monitoring role, reining in aggressive tax avoidance tactics that might expose the firm to overly high levels of related risk or costs. The preceding discussion leads to the following hypothesis:

**H1:** The association between the level of financial expertise on the audit committee and tax

avoidance is moderated by business-level strategy such that the relation is more strongly positive or weakly negative for defender-type firms than for prospector-type firms.

# 3. RESEARCH METHODS

# 3.1 Measuring Tax Planning

The tax literature has developed several proxies for tax avoidance (Hanlon and Heitzman 2010). Following recent studies (e.g., Cook et al. 2017; Higgins et al. 2015; Robinson et. al. 2012; Minnick and Noga 2010), we employ firms' cash effective tax rates to proxy for tax avoidance broadly and unrecognized tax benefits to capture risky tax planning. We also use total and permanent book-tax differences as tax avoidance measures following prior literature. All of our tax avoidance measures are defined in detail in Appendix A.

Our first measure of tax avoidance is the cash effective tax rate (*CASH\_ETR*). We define *CASH\_ETR* as cash taxes paid divided by pretax book income. *CASH\_ETR* reflects the extent to which managers view effective tax planning as the ability to minimize cash taxes paid. *CASH\_ETR* also reflects tax avoidance strategies that defer cash taxes paid to later periods as well as those that avoid tax entirely (Cook et al. 2017; Hanlon and Heitzman 2010; Minnick and Noga 2010). Lower values of *CASH\_ETR* represent higher levels of tax avoidance.

We use total book-tax differences (*BTD*) and permanent book-tax-difference (*PERMDIFF*) as our second and third proxies for firms' tax planning activities. The former contains information about both the broader impact of tax avoidance on income tax expense and more aggressive tax avoidance, including deferral (e.g., Chen et al. 2010; Wilson 2009). The latter captures the effects of ETR reconciliation items and represents non-conforming tax avoidance that does not reflect deferral strategies (Higgins et al. 2015; Hanlon and Heitzman 2010). Following Higgins et al. (2015), we define *BTD* as book income less estimated taxable income, scaled by lagged total assets and *PERMDIFF* as the difference between total book-tax differences and temporary book-tax-differences, scaled by lagged total assets. Prior research suggests that larger book-tax differences are associated with tax shelter activity (Wilson 2009) and higher proposed IRS audit adjustments (Mills 1998). Accordingly, larger values of *BTD* and *PERMDIFF* represent higher levels of tax planning.

We also build a measure based on uncertain tax benefits (Cook et al. 2017; Higgins et al. 2015). We use the firm's current-year additions to its uncertain tax benefits (i.e., additions to its tax contingency reserve) disclosed in the income tax footnote as a measure of the risk

and uncertainty associated with the firm's uncertain tax positions (*UTB*). Larger values of *UTB* represent higher levels of tax planning

We use industry-adjusted values for all of our tax avoidance measures to address potential industry effects. We calculate the mean of each tax avoidance measure by industry (two-digit SIC code) and year and subtract this mean from the raw measure to get the industry-adjusted value.

## 3.2 Measuring Firm Strategy

We rely on Bentley et al.'s (2013) and Higgins et al.'s (2015) discrete *STRATEGY* composite measure as our proxy for a firm's business strategy. Thus, we use the following six variables to construct our *STRATEGY* score: (1) the ratio of research and development to sales (RD5); (2) the ratio of employees to sales (EMP5); (3) the one-year percentage change in total sales (REV5); (4) the ratio of SG&A expenditures to sales (SGA5); (5) the standard deviation of total employees (6EMP5); and (6) the ratio of net PPE scaled by total assets (CAP5). See Appendix B for detailed descriptions of the six components of *STRATEGY*.

Consistent with Ittner et al. (1997), Bentley et al. (2013), and Higgins et al. (2015), we compute all variables using a rolling average of the respective yearly ratios over the five years prior to year t. We then rank each of the six variables by forming quintiles within each 2-digit SIC industry-year. Within each industry-year, observations in the top quintile receive a score of 5, those in the next quintile receive a score of 4, etc., and those in the lowest quintile are given a score of 1.8 For each firm-year, we sum the scores across the six

This article is protected by copyright. All rights reserved.

\_

<sup>&</sup>lt;sup>8</sup> The scoring for CAP5 is inverted because defenders are expected to have the highest capital intensity.

variables such that the maximum score a firm could receive is 30, and the minimum score a firm could receive is 6. Higher scores represent firms that follow a more prospector-type strategy, while lower scores represent firms that follow a more defender-type strategy. Following Bentley et al.'s (2013) and Higgins et al.'s (2015) classification scheme, we classify firms with *STRATEGY* scores ranging from 6 (the minimum) to 12 as defenders and firms with *STRATEGY* scores ranging from 24 to 30 (the maximum) as prospectors. Firms with *STRATEGY* scores ranging from 13 to 23 are analyzers.

## 3.3 Base Empirical Model

To investigate whether financial expertise on the audit committee and firm business strategy interact to influence firms' tax avoidance behavior, we first specify the following base regression model (subscripts suppressed):

TaxPlanning = 
$$\alpha + \beta_1\%FEDIR + \beta_2\%FEDIR*DEFENDER + \beta_3\%FEDIR*PROSPECTOR + \beta_4DEFENDER + \beta_5PROSPECTOR +$$
 (1)  $\gamma Controls + \varepsilon$ 

TaxPlanning is the dependent variable, representing the four tax avoidance measures discussed above (i.e., CASH\_ETR, BTD, PERMDIFF, and UTB). Firms that engage in more tax avoidance behavior will have lower values for CASH\_ETR, higher book-tax differences (BTD and PERMDIFF), and/or higher uncertain tax benefits (UTB). We estimate equation (1) separately for each of the four tax avoidance measures. %FEDIR is the proportion of independent financial expert directors on the audit committee and is one of the main independent variables of interest. DEFENDER and PROPSECTOR are indicator variables coded one if the STRATEGY score classification scheme discussed in the previous section

categorizes the firm as a defender or prospector, respectively. The standard errors are Huber-White robust standard errors.<sup>9</sup>

 $extit{\theta}_1$  captures the effect of financial expert audit committee members on tax avoidance for analyzers.  $extit{\theta}_2( extit{\theta}_3)$  captures the incremental effect of financial expert audit committee members on tax avoidance for defender-type (prospector-type) firms. While we do not make a prediction as to sign of the coefficient on %FEDIR itself ( $extit{\theta}_1$ ), we expect an incrementally positive (negative) association with tax avoidance for defenders (prospectors). Accordingly, a negative (positive) value for  $extit{\theta}_2( extit{\theta}_3)$  in the CASH\_ETR model and a positive (negative) value for  $extit{\theta}_2( extit{\theta}_3)$  in the BTD, PERMDIFF, and UTB models would provide initial support for our hypothesis.

To control for firm characteristics that may also relate to tax avoidance, we include a set of control variables (*Controls*). We draw these variables from prior tax avoidance literature. We include firm size (*SIZE*), research and development expense (*R&D*), capital expenditures (*CAPITAL*), the ratio of debt to total assets (*LEVERAGE*), the ratio of intangibles to total assets (*INTANG*), and the ratio of gross property, plant and equipment to total assets (*PPNE*) (Dyreng et al. 2010). We also add firm profitability (*ROA*) and net operating loss carryforwards (*NOL*) to control for firms' incentive to avoid income taxes (Chen et al. 2010; Rego 2003).

To further control for firms' incentives and opportunities to tax plan, we include an indicator variable identifying firms with foreign operations (*FOROP*). We also control for

This article is protected by copyright. All rights reserved.

-

<sup>&</sup>lt;sup>9</sup> Results are virtually identical if we use standard errors that are clustered by firm and year (Gow et al. 2010) or are unadjusted.

institutional ownership (*HELD\_PCT*) as an additional corporate governance mechanism. Moreover, we control for CEO total compensation (*CEOCOMP*) and stock-based compensation (*STKMIX*) because prior studies find that the equity incentives of executives play a significant role in firms' level of tax avoidance (Rego and Wilson 2012; Desai and Dharmapala 2006). We control for the level of firms' discretionary accruals (*DACC*) because Frank et al. (2009) find a positive association between financial reporting aggressiveness and tax avoidance. Finally, we include pre-tax cash operating flows (*PRETAXCF*) in our regression model.

In addition to general firm attributes, prior studies (e.g., Armstrong et al. 2015;
Richardson et al. 2013; Lanis and Richardson 2011) suggest a relation between board independence and tax avoidance. We control for certain board characteristics to ensure that %FEDIR is not capturing some dimension of board monitoring other than financial expertise. Prior studies generally suggest that a board's independence level is related to its monitoring effectiveness (e.g., Dechow et al. 1996). Thus, we control for board size (BSIZE) and board independence (%OUTDIR). We also control for the size (ACSIZE) and independence (100%INDAUD) of the audit committee because regulators view these attributes as important factors in the audit committee's ability to monitor effectively (Blue Ribbon Committee on Improving the Effectiveness of Corporate Audit Committees 1999; SOX 2002). All variables are defined in detail in Appendix A.

## 3.4 Defenders vs. Prospectors

After establishing initial results by estimating equation (1) using the full sample, we next focus our analyses on directly comparing defenders vs. prospectors for our main hypothesis tests. Using the sample of only defenders and prospectors, we specify the following regression model (subscripts suppressed):

TaxPlanning = 
$$\alpha + \theta_1$$
%FEDIR +  $\theta_2$ %FEDIR\*PROSPECTOR +  $\theta_3$ PROSPECTOR + (2)  
γControls +  $\varepsilon$ 

As discussed previously, *PROSPECTOR* is an indicator variable coded 1 for prospector-type firms, and 0 otherwise. Because we perform this regression analyses using only defender- and prospector-type firms, a 0 value for *PROSPECTOR* indicates defender-type firms. *TaxPlanning* represents the four tax avoidance measures discussed previously (*CASH\_ETR*, *BTD*, *PERMDIFF*, and *UTB*). *Controls* represents the same set of control variables as in the base regression. Consistent with equation (1), the standard errors are Huber-White robust standard errors.<sup>10</sup>

The coefficient on %FEDIR represents the association between financial expertise on the audit committee and tax avoidance for defender-type firms, and the interaction between %FEDIR and PROSPECTOR captures the degree to which this association differs for prospector-type firms. A positive coefficient on %FEDIR×PROSPECTOR in the CASH\_ETR

This article is protected by copyright. All rights reserved.

-

<sup>&</sup>lt;sup>10</sup> As with equation (1), results are virtually identical if we use standard errors that are clustered by firm and year (Gow et al. 2010) or are unadjusted.

model and a negative value on *%FEDIR×PROSPECTOR* in the *BTD*, *PERMDIFF*, and *UTB* models would be consistent with our hypothesis.

# 3.5 Data and Descriptive Statistics

Our sample selection process begins with all firms possessing available data from COMPUSTAT, covering the years from 2004 to 2012. We restrict the sample to the post-2003 period because the SOX provision regarding financial experts became effective in 2003. COMPUSTAT provides firm-specific financial information. We obtain board data from the GMI Ratings (previously Corporate Library) database. From GMI Ratings, we obtain each director's portfolio and manually check each audit committee member's credentials. We identify an audit committee member as a financial expert if s/he has experience as a Chief Financial Officer, Accounting Officer, Chief Accountant, Controller, Certified Public Accountant, Chartered Accountant, Financial Officer, Head of Accounting, Vice President of Accounting, Chief Executive Officer, and/or President (Badolato et al. 2014; SOX 2002).

We start with 47,879 firm-year observations from COMPUSTAT with sufficient data to calculate *STRATEGY* during the 2004 to 2012 period. We then exclude firm-year observations without board and/or audit committee data from GMI Ratings. This procedure

<sup>&</sup>lt;sup>11</sup> Although GMI Ratings provides a variable identifying financial expert directors on the audit committee (COMMITTEEAUDITFINANCIALEXPERT), this variable is missing in 2001, 2002, and 2003. It is also missing data for numerous observations in other years (i.e., 2009, 2010). As a result, we believe it is more accurate to identify audit committee financial experts by checking each director's portfolio directly.

<sup>&</sup>lt;sup>12</sup> We require five years of data (years t-5 to t-1) to compute each firm-year's value for *STRATEGY*. We also require an additional year's data to create lagged variables, expanding our data needs back to 1999.

reduces our sample to 19,709 observations. We then eliminate observations without Execucomp data necessary to calculate CEO equity incentives, reducing the sample to 12,942 observations. Finally, we exclude firm-year observations without sufficient data to calculate all tax avoidance and control variables. Our resultant final sample consists of 9,670 firm-year observations during the 2004-2012 period. To reduce the effects of extreme observations, we winsorize all continuous variables at the 1<sup>st</sup> and 99<sup>th</sup> percentiles.

#### -- INSERT TABLE 1, PANEL A ABOUT HERE -

Panel A of Table 1 reports descriptive statistics for the variables used in our base regression analysis. The mean (median) values for our tax avoidance variables are 0.027 (0.004), 0.017 (0.010), 0.016 (0.005), and 0.826 (0.330) for *CASH\_ETR*, *BTD*, *PERMDIFF*, and *UTB*, respectively. The statistics for our primary explanatory variables indicate that the mean proportion of independent financial experts on the audit committee is about 60 percent. Given that the mean audit committee size is 3.73 members, the average number of financial experts on the audit committee is 2.27, comparable with prior studies (e.g., Badolato et al. 2014). In addition, about 11.2 (5.6) percent of firm-years are classified as defenders (prospectors) according to the *STRATEGY* score discussed previously. The statistics for our other board and audit committee variables indicate that the average board in our sample is about 70 percent independent and that about 79 percent of sample firms have an audit committee comprised entirely of independent directors. It is also worth noting that the average firm in our sample is profitable and has positive pretax operating cash flows, with mean (median) *ROA* and *PRETAXCF* values of 0.108 (0.086) and 0.147

(0.129), respectively. Finally, firms in our sample are over 78 percent owned by institutions on average.

## -- INSERT TABLE 1, PANEL B ABOUT HERE

Panel B of Table 1 shows statistics for several board and audit committee characteristics, including financial expertise, broken out by strategy type (i.e., defenders and prospectors). As alluded to above, we categorize 1,080 firm-year observations as defenders and 541 firm-year observations as prospectors based on our classification scheme, consistent with Higgins et al. (2015), whose sample also consists of more defender (1,954) than prospector (1,242) observations. The statistics on board and audit committee characteristics show that prospectors have smaller boards and audit committees, less independent boards, and fewer financial experts on the audit committee than defenders.

# -- INSERT TABLE 2 ABOUT HERE --

Table 2 presents univariate Spearman correlations for our regression variables. The primary concern here is the potential for harmful collinearity among the independent variables. The highest correlation among the independent variables is 0.84 between *ROA* and *PRETAXCF*, which approaches a level sufficient to raise a concern. Two other correlations also exceed 0.6 (0.72 between *CAPITAL* and *PPNE* and -0.62 between *SIZE* and *CEOCOMP*). All other correlation coefficients are below 0.6. Overall, Table 2 suggests that collinearity is likely not a problem in our data. Nonetheless, we perform tests to ensure that our findings are not driven by the higher correlations listed above (discussed later in "Sensitivity Tests").

#### 4. EMPIRICIAL RESULTS

## 4.1 Baseline Regression Results – Full Sample

Table 3 presents the results for our base regression analysis, which includes the full sample. Across the four models, statistical significance is based on two-tailed p-values. The coefficient on *%FEDIR* represents the association between tax avoidance and financial expertise on the audit committee for analyzers only. It is positive in the *CASH\_ETR* model and negative in the other three, although not significantly so in any model.

The interaction terms relate to our hypothesis, illustrating the extent to which this association differs for defenders and prospectors relative to analyzers. *%FEDIR\*DEFENDER* is significantly negative (positive) at the 0.01 (0.05) level in the *CASH\_ETR* (*UTB*) model, and *%FEDIR\*PROSPECTOR* is significantly negative at least at the 0.10 level in the *BTD*, *PERMDIFF*, and *UTB* models. Further, while *%FEDIR\*DEFENDER* and *%FEDIR\*PROSPECTOR* are not statistically significant in the *BTD* and *PERMDIFF* and *CASH\_ETR* models, respectively, the signs of their coefficients are opposite of those for the significant interactions involving the opposing strategy type.

#### -- INSERT TABLE 3 ABOUT HERE --

Overall, these findings provide some initial (albeit somewhat weak) support for H1 and are generally consistent with the Adams and Ferreira (2007) view characterizing the board as serving dual roles as both monitors of and advisors to management. The more positive

association between financial expertise on the audit committee and tax avoidance for defenders (in the *CASH\_ETR* and *UTB* models) is consistent with these financial experts serving an advising role with respect to tax planning, encouraging firms of that strategy type to engage in more tax avoidance relative to other firms. Conversely, the more negative association between financial expertise on the audit committee and tax avoidance for prospectors (in the *BTD*, *PERMDIFF*, and *UTB* models) is consistent with these financial experts serving a monitoring role with respect to tax avoidance, encouraging firms of that strategy type to scale back on tax avoidance relative to other firms.

Results for the control variables reported in Table 3 are mixed but generally consistent with prior research. In accordance with the evidence in Higgins et al. (2015), *DEFENDER* is positive in the *CASH\_ETR* model and negative in the other three, although it is significant only in the *CASH\_ETR* and *PERMDIFF* models. Similarly, *PROSPECTOR* is negative in the *CASH\_ETR* model and positive in the other three, albeit significantly so only in the *PERMDIFF* and *UTB* models. With respect to the other board characteristics, *ACSIZE* is negatively related with tax avoidance in every model, significantly so in all but one (*UTB*). Board size (*BSIZE*), on the other hand, is negatively and significantly related with tax avoidance in the *PERMDIFF* model (p < 0.10) but is *positive* and significant (p < 0.01) in the *UTB* model.

\*\*COUTDIR\* and 100%INDAUD are significantly positive in the *UTB* model but insignificant in the other three.

Other notable statistics among the control variables include *R&D*, which is positively and significantly associated with tax avoidance in every model. Similarly, *ROA* and *LEVERAGE* 

(SIZE) are positively and significantly related to tax avoidance in all but the UTB (BTD) model. Several variables show mixed results. INTANG and FOROP are both significantly positive in the CASH\_ETR and UTB models and insignificant in the other two. Similarly, PPNE, HELD\_PCT, and STKMIX are significantly positive the BTD and PERMDIFF models but significantly negative in the UTB model. CEOCOMP and PRETAXCF are significant in every model but are directionally inconsistent across the tax avoidance measures.

## 4.2 Main Hypothesis Test Results - Defenders and Prospectors Only

The preceding baseline analysis uses the full sample, incorporating the entire strategy type continuum. For our main hypothesis tests, we re-estimate the regressions reported in Table 3 including only defender- and prospector-type firms to facilitate a stronger and more direct comparison between firms at opposite ends of the strategy continuum. The results for this analysis are reported in Table 4. Across the four models, statistical significance is again based on two-tailed p-values. The results for the control variables are generally similar to those reported in Table 3, although with some differences in significance levels.

%FEDIR, which now represents the association between tax avoidance and financial expertise on the audit committee for defenders, is significantly negative (p < 0.01) in the CASH\_ETR model and is positive in the other three models, albeit insignificantly so. For prospectors, %FEDIR\*PROSPECTOR is significantly positive (p < 0.01) in the CASH\_ETR model and significantly negative at least at the 0.05 level in the BTD, PERMDIFF, and UTB models.

Overall, the significant coefficients on *%FEDIR\*PROSPECTOR* in the predicted directions provide consistent support for H1. The results in Table 4 also consistently imply that, on

average, financial expert audit committee members rein in prospectors' (but not defenders') tax avoidance behavior and encourage defenders (but not prospectors) to engage in tax avoidance, at least with respect to cash taxes paid. These findings suggest that financial expert audit committee members vary how they approach their relationship with management according to the strategic profile of the firm, acting more as advisors with respect to defender-type firms and more as monitors with respect to prospector-type firms (Adams and Ferreira 2007).

#### -- INSERT TABLE 4 ABOUT HERE --

## 4.3 Accounting and Non-Accounting Financial Expertise on the Audit Committee

We next explore whether our main findings are driven by one or more types of financial expertise on the audit committee. Specifically, we expand on the results reported in Table 4 by decomposing %FEDIR into its accounting and non-accounting parts and testing for the extent to which either or both contribute to our findings. This test is motivated by the ongoing debate over what constitutes financial expertise for purposes of effective audit committee oversight. The SEC limited its original proposed definition of financial expertise for purposes of SOX to those with accounting-specific background and knowledge (e.g., CPAs, auditors, controllers, etc.). However, widespread criticism of such a narrow concept of financial expertise led to a much broader final definition that also includes those with

finance and relevant supervisory backgrounds (e.g., investment bankers, financial analysts, CEOs, presidents, etc.). <sup>13</sup>

The empirical evidence on whether non-accounting expertise contributes to the effectiveness of the audit committee is mixed. Krishnan and Visvanathan (2008) find a positive association between accounting expertise on the audit committee and accounting conservatism but no relation for non-accounting financial expertise. Conversely, Zhang et al. (2007) and Dhaliwal et al. (2010) find that both accounting and non-accounting financial experts contribute to the effectiveness of audit committee oversight in the areas of internal control weaknesses and accruals quality, respectively. As discussed previously, Robinson et al. (2012) find accounting expertise to be positively (negatively) related to levels of tax avoidance (risky tax avoidance). However, they also find a *positive* association between non-accounting financial expertise and risky tax avoidance, implying that non-accounting financial experts encourage managers to pursue aggressive and uncertain tax strategies on average.

To explore whether our main findings are driven by accounting financial experts, non-accounting financial experts, or both types on the audit committee, we re-perform the

<sup>&</sup>lt;sup>13</sup> Critics of the SEC's originally proposed definition argued that non-accounting financial experts often have considerable experience in carrying out due diligence with regard to forecasting future performance, developing business strategy, and coping with major corporate events. Accordingly, a narrow accounting-based definition of financial expertise is unnecessarily restrictive and limits the pool of qualified directors. For example, the American Association of Bank Directors argued that such a definition even disqualifies Alan Greenspan as a financial expert, and a Wall Street Journal article questioned whether Warren Buffet would meet the requirements as well (American Association of Bank Directors 2002).

regression analysis reported in Table 4 breaking %FEDIR into two components, accounting-specific expertise (%ACCDIR) and non-accounting-specific financial expertise (%NONACCDIR). We include both variables in the model and interact each of them with PROSPECTOR. As in Table 4, %ACCDIR and %NONACCDIR represent the association between financial expertise (accounting and non-accounting, respectively) and tax avoidance for defenders, and the interaction terms capture the degree to which those associations differ for prospectors.

#### -- INSERT TABLE 5 ABOUT HERE --

The results for this analysis are presented in Table 5; we omit the control variables for brevity. Looking at the association between financial expertise on the audit committee and tax avoidance for defenders, %ACCDIR (p < 0.10) and %NONACCDIR (p < 0.01) are both significantly negative in the CASH\_ETR model. %NONACCDIR is also significantly positive in the PERMDIFF model (p < 0.10). For prospectors, both %ACCDIR\*PROSPECTOR and %NONACCDIR\*PROSPECTOR are positive in the CASH\_ETR model and negative in the other three. However, while %NONACCDIR\*PROSPECTOR is highly significant in every model (p < 0.01), %ACCDIR\*PROSPECTOR is not significant in any model. These results suggest that while both types of financial expertise play a role in the tax planning process at some level, non-accounting financial expertise appears to generally have a more consistent impact across our tax avoidance measures.

In untabulated tests, we find that the results related to non-accounting financial expertise on the audit committee are primarily driven by audit committee members with

experience as a CEO (i.e., supervisory experience). Specifically, when we decompose *%NONACCDIR* into parts representing audit committee members with and without CEO experience, those with CEO experience show significant results more consistently than those without. Financial expert audit committee members with CEO experience show significant results in all four models, whereas those without CEO experience show significant results only in the *CASH\_ETR* and *PERMDIFF* models.

We perform Wald tests (untabulated) to directly compare the influence of accounting vs. non-accounting financial expert audit committee members on tax avoidance as implied by our Table 5 results. Examining differences in these associations separately for defenders (%ACCDIR vs. %NONACCDIR), the interaction terms themselves, and prospectors (the sums of the coefficients on %ACCDIR vs. %NONACCDIR and their respective interaction terms), we find little significant dissimilarity between accounting and non-accounting financial expert audit committee members. Specifically, %NONACCDIR is significantly more strongly positive than %ACCDIR in the PERMDIFF model (p < 0.01), and %NONACCDIR\*PROSPECTOR is significantly more strongly negative than %ACCDIR\*PROSPECTOR in the UTB model at marginal levels (p < 0.10). No other direct comparison between the influence of accounting vs. non-accounting expertise on the audit committee shows a significant difference.

Overall, these results suggest that both accounting and non-accounting financial expert audit committee members play a role with respect to firms' tax planning processes to some degree. Further, non-accounting financial experts (with supervisory experience) appear to significantly impact tax avoidance behavior more consistently than accounting financial

experts across our various tax measures, but their effects do not dominate those of accounting financial experts on average when compared directly.

# **4.4 Sensitivity Tests**

We perform multiple analyses to test the robustness of our main results. First, we consider the possibility that *PROSPECTOR* is endogenous in our tax avoidance models. Specifically, while our main analyses focus on the impact of firm strategy and its interaction with financial expertise on the audit committee on tax avoidance, it is possible that a firm's tax status/position might also partially drive some of the variables used to derive our firm strategy measure (e.g., research and development expense, number of employees, selling and administrative expenses, and capital expenditures).

We test for the endogeneity of *PROSPECTOR* using both the IVREG2 and IVENDOG commands in STATA. For purposes of these tests, we employ three instrumental variables for *PROSPECTOR*. The first is industry (proxied for using two-digit SIC code). Consistent with contingency theory, every organization is embedded in a network of external influences and relationships which can be labeled as its environment, such as product and labor market conditions, industry customs and practices, governmental regulations, and relationships with suppliers of financing and raw materials. Firms must match their business strategy to these environmental forces in order to succeed. For example, in turbulent environments, firms should pursue a strategy emphasizing innovation and differentiation, while more stable environments reward more conservative, low-cost approaches (Lee and Miller 1996). The second instrumental variable is the current age of the CEO. Prior research has shown

that older CEOs are more risk averse and are less likely to invest in R&D (Chowdury and Fink 2017; Kalyta 2009), which is critical to support a prospector strategy. Our use of this instrument stems from the likely role of the firm's top decision makers, who serve as the primary link between the organization and its environment, in setting business strategy (Miller and Snow 1978). The third instrument is the Herfindahl-Hirschman index based on total shareholders' equity, which captures industry concentration. Prior studies suggest that industry concentration affects firms' stock returns (Hou and Robinson 2006), informativeness of corporate disclosure policies (Ali et al. 2014), and innovation. Based on these findings, we conjecture that industry concentration likely affects a firm's business strategy as well. Results for these tests indicate no endogeneity for *PROSPECTOR* and suggest that OLS is consistent and appropriate for our main analyses. Specifically, across the four models, the endogeneity statistics produced by STATA (e.g., Wu-Hausman, Durbin-Wu-Hausman) are all insignificant, with p-values ranging from 0.28 to 0.54.

Our second set of sensitivity tests relates to the high correlations among some of our control variables discussed previously. First, we calculate variance inflation factors (VIFs) for the regression models reported in Table 4. Across all models and all variables, the highest VIF value is 5.73 (*PROSPECTOR* in the *BTD* model), indicating that our results are not likely

 $<sup>^{14}</sup>$  We test for the sufficiency and appropriateness of these instruments using the Kleibergen-Paap statistic for under-identification and the Cragg-Donald statistic for weak instruments based on the Stock and Yogo (2005) threshold levels for relative bias. In all cases, the Kleibergen-Paap statistic is highly significant (p < 0.01), indicating that our models are not under-identified. Further, the Cragg-Donald statistic exceeds the Stock and Yogo (2005) 10 percent relative bias threshold in every model, and the 5 percent relative bias threshold in three of four models. Based on these results, we conclude that our instrumental variables are relevant (i.e., representative of *PROSPECTOR*) and appropriate for our endogneneity tests.

affected by multicollinearity. Nonetheless, as an additional test to ensure that our results are not driven by the effects of collinearity, we re-estimate equation (2) three times, first omitting *PRETAXCF*, then (separately) omitting *PPNE*, and lastly (separately) omitting *CEOCOMP*. In all cases, our results for equation (2) remain qualitatively the same as those reported in Table 4.

## 4.5 Summary

Overall, our results suggest that financial expert audit committee members fulfill dual roles as advisors and monitors in the context of tax planning, and that their selection into these roles is at least partially driven by the business strategy of the firm. Our evidence of a negative association between *CASH\_ETR* and financial expertise on the audit committee for defenders is consistent with financial expert audit committee members serving more of an advising function for such firms, encouraging them to engage in more (cash) tax planning activity relative to other firms. On the other hand, our evidence is also consistent with financial experts on the audit committee serving more as monitors with respect to tax planning for prospector-type firms. For these firms, our results indicate a significantly more (and net) negative relation between tax avoidance and financial expertise on the audit committee overall. Additionally, our results suggest that both accounting and non-accounting financial experts on the audit committee influence tax planning at some level, although the impact of non-accounting financial experts appears to present more consistently than that of accounting financial experts across various tax measures. When

compared directly, however, the effects of non-accounting financial expertise do not dominate the effects of accounting financial expertise overall.

### 5. CONCLUDING REMARKS

This paper examines whether financial expert audit committee members tailor their approach to overseeing the corporate tax planning process according to the business strategy of the firm. Specifically, we investigate the extent to which such directors perform their dual roles as advisors and monitors (Adams and Ferreira 2007) with respect to tax avoidance differently across business strategy types. The management literature suggests that directors serving on various committees integrate their own expertise with their understanding of firm strategy in overseeing issues within the domain of the committee (Forbes and Milliken 1999; Zahra and Pearce 1989). We argue that this integration applies to financial expert audit committee members with respect to issues within the scope of audit committee oversight, including tax planning, such that they factor business strategy into their judgments and recommendations on such issues.

Following Miles et al. (1978) and Higgins et al. (2015), we classify firms into three broad strategy types, focusing primarily on (more risk averse) defenders and (more risk seeking) prospectors. Prior research suggests that managers of defenders are more likely to forgo tax avoidance opportunities that are potentially beneficial to shareholders while mangers of prospectors are more likely to be aggressive with their tax planning behavior, including that which is risky or uncertain (Higgins et al. 2015). Expanding on prior evidence (e.g., Higgins et al. 2015; Robinson et al 2012), we expect financial expert audit committee members to act

in more of an advising capacity in the tax planning process for defenders and in more of a monitoring capacity in the tax planning process for prospectors (relative to one another).

Our results are consistent with our prediction. We find a positive association between the proportion of independent financial experts on the audit committee and tax avoidance for defenders, at least with respect the *cash* tax planning. We also find a consistently more (and net) negative association between the proportion of independent financial experts on the audit committee and tax avoidance for prospectors, relative to defenders. Further, results of supplemental tests indicate that both accounting and non-accounting financial experts on the audit committee contribute to our main findings at some level, although the effects of non-accounting financial experts present more consistently across our analyses.

Overall, our results suggest that financial expert audit committee members play more of an advising role for defender-type firms with respect to tax planning broadly and more of a monitoring role for prospector-type firms. These findings build upon prior studies as we show that financial expert audit committee members alter the manner in which they perform their advisory and monitoring roles around tax planning issues according to the firm's business strategy. Thus, our study helps to provide a richer understanding of how board members approach oversight of the corporate tax planning process. Our evidence also extends Higgins et al. (2015) by documenting that certain corporate governance mechanisms influence managerial decision making differentially across business strategy types. One potential implication of our findings is that future studies examining the impact of board characteristics on firms' operational or reporting behavior may benefit from incorporating business strategy into their analyses.

#### **REFERENCES**

- Adams, R., and D. Ferreira. 2007. A theory of friendly boards. *Journal of Finance*, 62: 217-250.
- Ali, A., S. Klasa, and E. Yeung. 2014. Industry concentration and corporate disclosure policy. *Journal of Accounting and Economics* 58(2): 240-264.
- American Association of Bank Directors. 2002. Letter to the SEC RE: The Sarbanes-Oxley Act of 2002.
- Armstrong, C., J.L. Blouin, A.D. Jagolinzer, and D.F. Larcker. 2015. Corporate governance, incentives, and tax avoidance. *Journal of Accounting and Economics*, 60(1): 1-17.
- Badolato, P. G., D. C. Donelson, M. Ege. 2014. Audit committee financial expertise and earnings management: The role of status. *Journal of Accounting and Economics*, 58(2): 208-230.
- Baginski, S. P., Lorek, K. S., Willinger, G. L., & Branson, B. C. 1999. The relationship between economic characteristics and alternative annual earnings persistence measures. *The Accounting Review*, 74: 105-120.
- Baysinger, B.D., and R. E. Hoskisson. 1990. The composition of boards of directors and strategic control: Effects of corporate strategy. *Academy of Management Review*, 15(1): 72-87.
- Baysinger, B., and H. Butler. 1985. Corporate governance and the board of directors:

  Performance effects of changes in board composition. *Journal of Law, Economics, and Organizations*, 1: 101-124.
- Baum, C.F., M.E. Schaffer, and S. Stillman. "Enhanced routines for instrumental variables/GMM estimation and testing." *Stata Journal* 7.4 (2007): 465-506.
- Bentley, K.A., T.C. Omer, and N.Y. Sharp. 2013. Business strategy, audit fees and financial reporting irresagularities. *Contemporary Accounting Research*, 30 (2): 780-817.
- Blue Ribbon Committee on Improving the Effectiveness of Corporate Audit Cmmittees.

  1999. Report and recommendations of the Blue Ribbon Committee on Improving the Effectiveness of Corporate Audit Committees. New York: New York Stock Exchange and the National Association of Securities Dealers.
- Byrd, J.W., and K.A. Hickman. 1992. Do outside directors monitor managers?: Evidence from tender offer bids. *Journal of Financial Economics*, 32 (2): 195-221.

- Carpenter, R. N. 1988. Cooperative governance, part II. Directors' responsibilities. *Directors & Boards*, 29(3): 3-6.
- Chan, K., P. Mo, and A. Zhou. 2013. Government ownership, corporate governance and tax aggressiveness: Evidence from China. *Accounting and Finance*, 53: 1029-1051.
- Chen, S., X. Chen, Q. Cheng, and T. Shevlin. 2010. Are family firms more tax aggressive than non-family firms? *Journal of Financial Economics*, 95 (1): 41-61.
- Chowdhury, J., and J. Fink. 2017. How does CEO age affect firm risk? *Asia-Pacific Journal of Financial Studies*, 46: 381-412.
- Cook, K., W. Moser, and T. Omer. 2017. Tax avoidance and *ex ante* cost of capital. *Journal of Business Finance & Accounting*, 44 (7-8): 1109-1136.
- Cotter, J.F., A. Shivdasani, and M. Zenner. 1997. Do independent directors enhance target shareholder wealth during tender offers? *Journal of Financial Economics*, 43 (2): 195-218.
- Dechow, P. M., R.G. Sloan, and A.P. Sweeney. 1996. Causes and consequences of earnings manipulation: An analysis of firms subject to enforcement actions by the SEC. *Contemporary Accounting Research*, 13 (1): 1-36.
- Deloitte LLP. 2011. Hot Topics Taxes: What the audit committee should know. Available at: <a href="http://www.iasplus.com/en-us/publications/us/hot-topics/2011/october">http://www.iasplus.com/en-us/publications/us/hot-topics/2011/october</a>.
- Deloitte LLP. 2013. Audit Committee Brief: Top issues for audit committees in 2014.

  Available at: <a href="http://www.iasplus.com/en-us/publications/us/acb/2013/november-december">http://www.iasplus.com/en-us/publications/us/acb/2013/november-december</a>.
- Deloitte LLP. 2014. Audit Committee Brief: What's on your agenda for 2015. Available at: http://www.iasplus.com/en-us/publications/us/acb/2014/nov-dec.
- Desai, M.A., and D. Dharmapala. 2006. Corporate tax avoidance and high-powered incentives. *Journal of Financial Economics*, 79 (1): 145-179.
- Dhaliwal, D.A.N., V.I.C. Naiker, and F. Navissi. 2010. The association between accruals quality and the characteristics of accounting experts and mix of expertise on audit committees. *Contemporary Accounting Research*, 27 (3): 787-827.
- Dyreng, S.D., M. Hanlon, and E.L. Maydew. 2010. The effects of executives on corporate tax avoidance. *The Accounting Review*, 85 (4): 1163-1189.
- Eisenhardt, K.M. 1985. Control: Organizational and economic approaches. *Management Science*, 31(2): 134-149.

- Fama, E. F., and M.C. Jensen. 1983. Separation of ownership and control. *Journal of Law and Economics*, 26 (2): 301-325.
- Farber, D.B. 2005. Restoring trust after fraud: Does corporate governance matter? *The Accounting Review*, 80 (2): 539-561.
- Forbes, D. P., and F. J. Milliken.1999. Cognition and corporate governance: Understanding boards of directors as strategic decision-making groups. *Academy of Management Review*, 24(3): 489-505.
- Frank, M.M., L.J. Lynch, and S.O. Rego. 2009. Tax reporting aggressiveness and its relation to aggressive financial reporting. *The Accounting Review*, 84 (2): 467-496.
- Gow, I., G. Ormazabal, and D. Taylor. 2010. Correcting for cross-sectional and time-series dependence in accounting research. *The Accounting Review*, 85 (2): 483-512.
- Gupta, A.K. 1987. SBU strategies, corporate-SBU relations, and SBU effectiveness in strategy implementation. *Academy of Management Journal*, 30(3): 477-500.
- Hambrick, D. C., and C.C. Snow. C. 1989. Strategic reward systems. In C. C. Snow (Ed.), Strategy, organization design, and human resource managaement (pp. 333-367). Greenwich, CT: JAI Press.
- Hanlon, M., and S. Heitzman. 2010. A review of tax research. *Journal of Accounting and Economics*, 50 (2):127-178.
- Higgins, D., T. Omer, and J. Phillips. 2015. The influence of a firm's business strategy on its tax aggressiveness. *Contemporary Accounting Research*, 32 (2):674-702.
- Hoskisson, R.E., and M.A. Hitt. 1988. Strategic control systems and relative R&D investment in large multiproduct firms. *Strategic Management Journal*, 9: 506-621.
- Hou, K., and D.T. Robinson. 2006. Industry concentration and average stock returns. *The Journal of Finance* 61(4): 1927-1956.
- Huseynov, F., and B. Klamm. 2012. Tax avoidance, tax management and corporate social responsibility. *Journal of Corporate Finance*, 18 (4): 804-827.
- Ittner, C.D., D.F. Larcker, and M.V. Rajan. 1997. The choice of performance measures in annual bonus contracts. *The Accounting Review*, 72 (2): 231-255.
- Jemison, D., and P. Oakley. 1983. Corporate governance in mutual insurance companies. *Journal of Business Research*, 11 (4): 501-522.
- Kalyta, P. 2009. Accounting discretion, horizon problem, and CEO retirement benefits, *Accounting Review*, 84: 1553–1573.

- Khurana, I., and W. Moser. 2013. Institutional shareholders' investment horizons and tax avoidance. *Journal of the American Taxation Association*, 35(1): 111-134.
- Klein, A. 2002. Audit committee, board of director characteristics, and earnings management. *Journal of the Accounting and Economics*, 33: 375-400.
- Kothari, S.P., A. Leone, and C. Wasley. 2005. Performance matched discretionary accrual measures. *Journal of Accounting and Economics*, 39 (1): 163-197.
- Krishnan, and G. Visvanathan. 2008. Does the SOX definition of an accounting expert matter? The association between audit committee directors' accounting expertise and accounting conservatism. *Contemporary Accounting Research*, 25 (3): 827-857.
- KPMG LLP. 2012. Is governance keeping pace? Available at:

  <a href="http://www.kpmg.com/Global/en/IssuesAndInsights/ArticlesPublications/Documents/">http://www.kpmg.com/Global/en/IssuesAndInsights/ArticlesPublications/Documents/</a>

  s/kpmg-2012-audit-committee-issues-conference.pdf.
- KPMG LLP. 2015. 2015 Global Audit Committee Survey. Available at:

  <a href="https://boardleadership.kpmg.us/relevant-topics/articles/2015/01/2015-global-audit-committee-survey.html">https://boardleadership.kpmg.us/relevant-topics/articles/2015/01/2015-global-audit-committee-survey.html</a>.
- KPMG LLP. 2017. Is everything under control? 2017 Global Audit Committee Pulse Survey. Available at: <a href="https://boardleadership.kpmg.us/relevant-topics/articles/2017/01/2017-global-audit-committee-pulse-survey.html">https://boardleadership.kpmg.us/relevant-topics/articles/2017/01/2017-global-audit-committee-pulse-survey.html</a>.
- Lanis, R., and G. Richardson. 2011. The effect of board of director composition on corporate tax aggressiveness. *Journal of Accounting and Public Policy*, 30 (1): 50-70.
- Lee, J.. and D. Miller. 1996. Strategy, environment and performance in two technological contexts: Contingency theory in Korea. *Organizational Studies*, 17 (5): 729-750.
- Miles, R.E., C.C. Snow, A.D. Meyer, and H.J. Coleman. 1978. Organizational strategy, structure, and process. *Academy of Management Review*, 3(3): 546-562.
- Miller, D. 1987. The structural and environmental correlates of business strategy. *Strategic Management Journal*, 8(1): 55-76.
- Mills, L.F. 1998. Book-tax differences and Internal Revenue Service adjustments. *Journal of Accounting Research*, 36 (2): 343-356.
- Minnick, K., and T. Noga. 2010. Do corporate governance characteristics influence tax management? *Journal of Corporate Finance*, 16 (5): 703-718.
- Mizruchi, M. S. 1983. Who controls whom? An explanation of the relation between management and boards of directors in large American corporations. *Academy of Management Review*, 8(3): 426-435.

- Moore, J., S. Suh, and E. Werner. 2017. Dual entrenchment: Classified boards and family firms. *Journal of Business Research*, 79: 161-172.
- Pfeffer, J., and G. Salancik. 1978. *The external control of organizations: A resource dependence perspective.* New York: Harper & Row.
- Phillips, J.D. 2003. Corporate tax-planning effectiveness: The role of compensation-based incentives. *The Accounting Review*, 78 (3): 847-874.
- Public Oversight Board. 1993. In the public interest: Issues confronting the accounting profession. *Stamford, CT: American Institute of Certified Public Accountants*.
- Rego, S.O. 2003. Tax-avoidance activities of US multinational corporations. *Contemporary Accounting Research*, 20 (4): 805-833.
- Rego, S.O., and R. Wilson. 2012. Equity risk incentives and corporate tax aggressiveness. *Journal of Accounting Research*, 50 (3): 775-810.
- Richardson, G., G. Taylor, and R. Lanis, 2013. The impact of board of director oversight characteristics on corporate tax aggressiveness: An empirical analysis. *Journal of Accounting and Public Policy*, 32(3): 68-88.
- Robinson, J., Y. Xue, and M. Zhang. 2012. Tax planning and financial expertise in the audit committee. Working Paper.
- Rogers, J. L., and Stocken, P. C. 2005. Credibility of management forecasts. *The Accounting Review*, 80(4): 1233-1260.
- Sarbanes-Oxley Act of 2002 (SOX). 2002. Public Law No. 107-204, 116 Stat. 745, Sec. 1-1107.
- Slater, S.F., and E.M. Olson. 2000. Strategy type and performance: The influence of sales force management. *Strategic Management Journal*, 21(8): 813-829.
- Slemrod, J. 2004. The economics of corporate tax selfishness. *National Tax Journal*, 57(4): 877-899.
- Steijvers, T., and M. Niskanen. 2014. Tax aggressiveness in private family firms: An agency perspective. *Journal of Family Business Strategy*, 5: 347-357.
- Stock J, and M. Yogo. 2005. Testing for weak instruments in linear IV regression. In: Andrews DWK Identification and Inference for Econometric Models. New York: Cambridge University Press: 80-108.
- Weisbach, M.S. 1988. Outside directors and CEO turnover. *Journal of Financial Economics*, 20: 431-460.
- Wilson, R.J. 2009. An examination of corporate tax shelter participants. *The Accounting Review*, 84(3): 969-999.

- Xie, B., W. Davidson III, and P. DaDalt. 2003. Earnings management and corporate governance: The role of the board and the audit committee. *Journal of Corporate Finance*, 9: 295-316.
- Zahra, S.A., and J.A. Pearce II. 1989 Boards of directors and corporate financial performance: A review and integrative model. *Journal of Management*, 15(2): 291-334.
- Zahra, S.A., and J.A. Pearce, II. 1990. Research evidence on the Miles-Snow typology. *Journal of Management*, 16(4): 751-786.
- Zhang, Y., J. Zhou, and N. Zhou. 2007. Audit committee quality, auditor independence, and internal control weaknesses. *Journal of Accounting and Public Policy*, 26(3): 300-327.

## **Appendix A: Variable Definitions**

Tax Avoidance Measures	
CASH_ETR	Industry-adjusted cash effective tax rate. Cash ETR is defined as cash taxes paid (TXPD) divided by pretax book income (PI). Cash ETRs with negative denominators are deleted. The remaining non-missing ETRs are winsorized (reset) so that the largest observation is equal to 1 and the smallest is equal to 0. The industry mean of cash ETR
	is calculated by year and industry (two-digit SIC code). Industry-adjusted cash ETR is calculated as cash ETR less the industry mean.
BTD	Industry-adjusted total book tax differences. Total book-tax differences are defined as pretax income less estimated taxable income, scaled by lagged assets (AT). Pretax book income is defined as pretax income (PI). Estimated taxable income is defined as the sum of current federal tax expense (TXFED) and current foreign tax expense (TXFO) divided by the top U.S. statutory tax rate. If current federal tax expense is missing, then we calculate it as total tax expense (TXT) minus deferred tax expense (TXDI), state income taxes (TXS), and current

foreign tax expense (TXFO). The industry mean of total book-tax differences is calculated by year and industry (two-digit SIC code). Industry-adjusted BTD is calculated as total book-tax differences less the industry mean.
Industry-adjusted permanent book-tax differences.  Permanent book-tax differences are computed as the difference between total book-tax differences and temporary book-tax differences, scaled by beginning of the year total assets (AT). Total book-tax differences are defined above. Temporary book-tax differences are defined as deferred tax expense (TXDI) divided by the top U.S. statutory tax rate. The industry mean of permanent BTDs is calculated by year and industry (two-digit SIC code). Industry-adjusted PERMDIFF is calculated as permanent book-tax differences less the industry mean.
Industry-adjusted unrecognized tax benefits, defined as the log value of one plus the increase in current uncertain tax positions (TXTUBPOSINC). The industry mean of UTB is calculated by year and industry (two-digit SIC code). Industry-adjusted UTB is calculated as unrecognized tax benefits less the industry mean.

Board Measures	
%FEDIR	Percentage of independent financial experts on the audit committee.
%ACCDIR	Percentage of independent accounting experts on the audit committee. A director is defined as an accounting expert if he or she has work experience as a certified public accountant, auditor, CFO, financial comptroller, financial controller, or accounting officer.
%NONACCDIR	Percentage of independent non-accounting financial experts on the audit committee. A director is defined as

	and the financial area of the control of the second state of the s
	a non-accounting financial expert if he or she does not have work experience as a certified public accountant, auditor, CFO, financial comptroller, financial controller, or accounting officer but does have work experience as an investment banker, financial analyst, or any other financial management role, or experience supervising the preparation of financial statements (e.g., CEO or company president).
BSIZE	Log value of the number of directors on the board.
%OUTDIR	Percentage of independent directors on the board.
ACSIZE	Log value of the number of audit committee members.
100%INDAUD	Indicator variable coded 1 if the audit committee is composed 100% of independent directors.
STRATEGY Components	
RD5	Five year rolling average (year t-5 through year t-1) of the yearly ratio of research and development expense (XRD) to beginning of year total sales (SALE).
EMP5	Five year rolling average (year t-5 through year t-1) of the yearly ratio of the total number of employees (EMP) to beginning of year total sales (SALE).
REV5	Average annual percentage change in total sales (SALE) computed over a rolling five year period (year t-5 through year t-1).
SGA5	Five year rolling average (year t-5 through year t-1) of the yearly ratio of SG&A expense (XSGA) to beginning of year total sales (SALE).
бЕМР5	Standard deviation of the total number of employees (EMP) computed over a rolling prior five year period (year t-5 through year t-1).
CAP5	Five year rolling average (year t-5 through year t-1) of the yearly ratio of net property, plant, and equipment

(PPENT) to beginning of year total assets (AT).

Control variables	
SIZE	Log value of lagged total assets (AT).
ROA	Pre-tax income (PI), scaled by lagged total assets (AT).
R&D	Research and development expense (XRD), scaled by lagged total assets (AT). When XRD is missing, we set the value to 0.
CAPITAL	Capital expenditures (CAPX) scaled by lagged total assets (AT).
LEVERAGE	Sum of long-term debt (DLTT) and long-term debt in current liabilities (DLC), scaled by lagged total assets (AT).
INTANG	Intangible assets (INTAN), scaled by lagged total assets (AT).
PPNE	Gross property, plant and equipment (PPEGT), scaled by lagged total assets (AT).
NOL	Indicator variable coded 1 for firms with non-missing and non-zero values for tax loss carryforward (TLCF).
FOROP	Indicator variable coded 1 for firms with non-missing and non-zero values for pretax foreign income (PIFO).
HELD_PCT	Percent of outstanding shares held by institutions (from Thomson-Reuters 13f Holdings).
DACC	Adjusted discretionary accruals based on the performance-adjusted modified cross-sectional Jones model.
CEOCOMP	CEO total compensation (Salary + Bonus + Other Annual + Restricted Stock Grants + LTIP Payouts + All Other + Value

	of Options Exercised), scaled by lagged total assets (AT).
STKMIX	CEO stock-based compensation, defined as the value of overall stock based compensation (OPTION_AWARDS_BLK_VALUE, RSTKGRNT) as a fraction of total compensation (OPTION_AWARDS_BLK_VALUE, RSTKGRNT, SALARY, and BONUS).
PRETAXCF	Pretax cash flows, defined as cash flows from operating activities (OANCF) plus taxes paid (TXPD), scaled by lagged total assets (AT).

# **APPENDIX B: Business Strategy Characteristics**

	Prospector	Analyzer	Defender
Research and Development (RD5)	Extensive R&D to identify new products and market opportunities	Extensive R&D to identify new products and market opportunities	Minimal R&D, which is usually related to existing products
Employees (EMP5)	More employees per dollar of sales	Moderate employees per dollar of sales	Fewer employees per dollar of sales because defenders focus on organizational efficiency

Growth (REV5)

Growth occurs in spurts

)		through product and market development.	both market penetration and product and market development	incremental growth and advances in productivity
	Marketing (SGA5)	Strong focus on marketing.	Strong focus on marketing in innovative sector.	Strong emphasis on financial and production functions and less on
, 	Employee	Higher employee	Madarata amplayas	marketing.
	Employee turnover (σEMP5)	Higher employee turnover focusing on shorter employee tenure	Moderate employee turnover	Low employee turnover focusing on lengthy employee tenure and promotion from within the firm
	Capital	Low degree of mechanization and	Moderate degree of	High degree of
)	Intensity (CAP5)	routinization to avoid a	mechanization and	mechanization and
)	(6, 11 3)	lengthy commitment to a single technological	routinization while	routinization focusing on a single core-efficient
) .		process	remaining flexible	technology
			enough to pursue new business activities	
			DUSINESS ACTIVITIES	

Steady growth through

Cautious and

				Tab	ole:	l¤							
			D	escriptiv	e-S	tatistics¤							
Panel:A: Full-Sample¤	92	٩	ž	9a	90	¥	¥	°⊭	٩	°¤	¥	¥	9
Ħ	9:	%	92	9a	90	<b>%</b>	%	<b>9</b> α	92	<b>9</b> a	90	<b>9</b> α	٩
Variablesa	9:	·N·12	90	Means	90	Std. Devæ	ъ	25%¤	90	Mediano	90	75%s	1
Continuous-	٩		9	%	٩		я		9:	_	٩		9
CASH_ETR¤	٩	9,670≉					94	-0.093¤	9	0.004¤	¥	0.094	9
BTD¤		7,758≉					94	-0.008¤	9	0.010¤	¥	0.034	9
PERMDIFF¤		7,741≉					94	-0.004¤	٩	0.005¤	ч	0.022	9
UTB¤		5,147≴					94	-0.151¤	٩	0.330¤	ч	1.597	9
#∙of·FEDIR¤		9,670≉					94	1¤	٩	2¤	Ħ	3⊧	9
%FEDIR¤	9,	9,670≴	¥	0.606¤	٦	0.290¤	94	0.333¤	٩	0.667¤	Ħ	0.800=	9
STRATEGY¤	٩	9,670≮	¥	17.099¤	٦	3.810¤	94	14¤	٩	17¤	Ħ	20⊧	9
SIZE¤	٩	9,670≮	٩	7.839¤	٩	1.657¤	94	6.636¤	٩	7.691¤	Ħ	8.894	9
ROA¤	٩	9,670≮	٩	0.108¤	٩	0.090¤	94	0.042¤	٩	0.086¤	Ħ	0.149	9
R&D¤	9:	9,670≮	¥	0.024¤	۴	0.047⊭	94	0.000¤	٩	0.000≒	Ħ	0.027	9
CAPITAL¤	9:	9,670≮	¥	0.049¤	۴	0.058¤	94	0.014¤	٩	0.032¤	Ħ	0.063=	9
LEVERAGE¤	9:	9,670≮	¥	0.228¤	۴	0.210¤	94	0.053¤	٩	0.194¤	Ħ	0.336=	9
INTANG¤	9:	9,670≮	¥	0.203¤	۴	0.226¤	94	0.019¤	٩	0.122≒	Ħ	0.318=	9
PPNE¤		9,670≮					94	0.144¤	٩	0.371≒	Ħ	0.759=	9
HELD PCT¤	9.	9,670≉	٩	0.785¤	٩	0.205¤	9	0.667¤	٩	0.812¤	¥	0.923=	9
DACC¤	ą.	9,670≮	4	-0.001¤	٦	1.589¤	9	-0.040¤	۴	0.011¤	¥	0.099	9
CEOCOMP#	ą.	9,670≮	¥	2.835¤	۴	3.494¤	94	0.630¤	4	1.670¤	¥	3.627	9
STKMIX¤	ą	9,670≮	¥	0.562¤	۴	0.300¤	94	0.401¤	4	0.662¤	¥	0.797	9
PRETAXCF¤	۰	9,670≉	¥	0.147¤	۴	0.104¤	94	0.075¤	4	0.129♯	¥	0.198	9
#-of-Directors¤	-3	9,670≉					9	8¤	9	9⊭	¥	115	-9
#81ZE¤		9,670¢					9	2.079⊭	9	2.197⊭	4	2.398=	9
%OUTDIR¤	Ž						94	0.600¤	9	0.714¤	91	0.833=	-9
%00 IDIK∺ #-of-Audit-Committee-Mbrs¤	<del>!</del> 3		- 1								헠		
#-01-Audit-Committee:Wors= ACSIZE¤	ą	9.670≉	- 1		! i			_		1.609¤			
	ž		9	9	4						91		
Categorical <sup>®</sup>	- 3	9.670≉	4	0.112	4	0.315¤	9				4		- 3
DEFENDER¤	- 5	9,670≉	- 1		i !						4	_	-3
PROSPECTOR¤	- 5	9,670 ≉	4	0.6189	4	0.486¤					4	_	- 3
NOL¤		9,670 ≉	-	0.5538	ą	0.497¤			4	Ĵ			,
FOROP		9,670≉			: 1				4	<u>9</u>			- 9
100%INDAUD¤	- 3												
9≒	35	°¤	95	°#	%:	% :	93	°≒	35	≌	34	°≒	9

		Table	1				
	Descr	iptive Sta	tistics	(cont.)			
Panel B: Defenders and							
<u>Prospectors</u>							
		Mea		Std.		Medi	
Variables	N 	n	_	Dev.	25%	an	75%
				DEFENDE	R = 1		
# of Directors		10.14	**	2.486	8	10	12
	1,08 0	1	*				
%OUTDIR		0.724	**	0.158	0.62	0.750	0.87
	1,08 0		*		5		5
of Audit Committee Mbrs		3.897	**	1.094	3	4	5
	1,08 0		*				
% Independent Audit		0.926		0.166	1	1	1
Committee Mbrs	1,08 0						
# of FEDIR		2.419	**	1.288	1	2	3
	1,08 0		*				
%FEDIR		0.618	**	0.274	0.40	0.667	0.80
	1,08 0		*		0		0
# of ACCDIR		0.450		0.611	0	0	1
	1,08 0						
%ACCDIR	1,08	0.122		0.171	0	0	0.25 0

	0						
# of NONACCDIR	1,08 0	1.969	**	1.268	1	2	3
%NONACCDIR	1,08 0	0.496	**	0.282	0.33	0.500	0.66 7
				PROSPECT	OR = 1		
# of Directors	541	9.190		2.655	7	9	11
%OUTDIR	541	0.686		0.164	0.57 1	0.714	0.83
# of Audit Committee Mbrs	541	3.580		0.915	3	3	4
% Independent Audit Committee Mbrs	541	0.919		0.169	1	1	1
# of FEDIR	541	2.033		1.187	1	2	3
%FEDIR	541	0.566		0.299	0.33	0.667	0.75 0
# of ACCDIR	541	0.458		0.597	0	0	1
%ACCDIR	541	0.137		0.184	0	0	0.33
# of NONACCDIR	541	1.575		1.199	1	1	2
				0.300	0.25	0.333	0.66

Notes to Table 1:

\*\*\* indicates a statistically significant difference between defenders and prospectors at the 1% level.

CASH\_ETR is cash taxes paid divided by pretax book income. Cash ETRs with negative denominators are deleted. The remaining non-missing ETRs are winsorized (reset) so that the largest observation is equal to 1 and the smallest is equal to 0. This variable is industry-adjusted by subtracting the industry mean (by year and two-digit SIC code). BTD is total book-tax differences, defined as pretax income less estimated taxable income, scaled by lagged assets. Estimated taxable income is the sum of current federal tax expense and current foreign tax expense divided by the top U.S. statutory tax rate. If current federal tax expense is missing, then we calculate it as total tax expense minus deferred tax expense, state income taxes, and current foreign tax expense. This variable is industryadjusted by subtracting the industry mean (by year and two-digit SIC code). PERMDIFF is the difference between total book-tax differences (defined as in BTD) and temporary book-tax differences, scaled by beginning of the year total assets. Temporary book-tax differences are deferred tax expense divided by the top U.S. statutory tax rate. This variable is industry-adjusted by subtracting the industry mean (by year and two-digit SIC code). UTB is the log value of one plus the increase in current uncertain tax positions, industry-adjusted by subtracting the industry mean (by year and two-digit SIC code). # of FEDIR is the number of independent financial experts on the audit committee. %FEDIR is the percentage of independent financial experts on the audit committee. STRATEGY is the firm strategy score for firm i in year t, measured following Bentley et al. (2013) and Higgins et al. (2015). The maximum (minimum) value for this variable is 30 (6). SIZE is the log value of lagged total assets. ROA is pre-tax income, scaled by lagged total assets. R&D is research and development expense, scaled by lagged total assets. When XRD is missing, we set the value to 0. CAPITAL is capital expenditures scaled by lagged total assets. LEVERAGE is the sum of long-term debt and long-term debt in current liabilities, scaled by lagged total assets. INTANG is intangible assets, scaled by lagged total assets. PPNE is gross property, plant and equipment, scaled by lagged total assets. HELD\_PCT is the percentage of outstanding shares held by institutions (from Thomson-Reuters 13f Holdings). DACC is adjusted discretionary accruals based on the performance-adjusted modified cross-sectional Jones model, measured following Frank et al. (2009) and Kothari et al. (2005). CEOCOMP is CEO total compensation, scaled by lagged total assets. STKMIX is CEO stockbased compensation as a fraction of total compensation. PRETAXCF is pretax cash flows, defined as cash flows from operating activities plus taxes paid, scaled by lagged total assets. # of Directors is number of directors on the board. BSIZE is the log value of the number of directors on the board. %OUTDIR is the percentage of independent directors on the board. # of Audit Committee Mbrs is the number of audit committee members. ACSIZE is the log value of the number of audit committee members. DEFENDER is an indicator variable coded 1 if the STRATEGY score for firm i in year t ranges from 6 (the minimum) to 12, 0 otherwise. PROSPECTOR is an indicator variable coded 1 if the STRATEGY score for firm i in year t ranges from 24 to 30 (the maximum), 0 otherwise. NOL is an indicator variable coded 1 for firm-years with non-missing and non-zero values for tax loss carryforward, 0 otherwise. FOROP is an indicator variable coded 1 for firm-years with non-missing and non-zero values for pretax foreign income, 0 otherwise. 100%INDAUD is an indicator variable coded 1 if the audit committee is composed 100% of independent directors, 0 otherwise. % Independent Audit Committee Mbrs is the percentage of independent audit committee members. #

of ACCDIR is the number of independent accounting experts on the audit committee. A director is defined as an accounting expert if he/she has work experience as a certified public accountant, auditor, CFO, financial comptroller, financial controller, or accounting officer. %ACCDIR is the percentage of independent accounting experts on the audit committee. # of NONACCDIR is the number of independent non-accounting financial experts on the audit committee. A director is defined as a non-accounting financial expert if he or she does not have work experience as a certified public accountant, auditor, CFO, financial comptroller, financial controller, or accounting officer but does have work experience as an investment banker, financial analyst, or any other financial management role, or experience supervising the preparation of financial statements (e.g., CEO or company president). %NONACCDIR is the percentage of independent non-accounting financial experts on the audit committee. See Appendix A for variable definitions.

											Ta	ble∙2¤													
												ion-Mo													
6	1=	2=	3=	40	50	6=	7=	80	90	10=	11=	12=	13=	14=	150	16=	17=	18=	19=	20=	21=	22=	23=	24=	25=
:-CASH_ETRP	1.000			_		_		_	_		-	_		_			_				_		-		
Bille		1.000	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	-	
PERMODES		0.535			-	-	~	-		~	-	-	~	-	-	~	~	-		~	-	-	-	~	
C-01Bs	-0.01=	0.03=	0.062	1.00=	-	-	-	_	-	-	-	-	_	~	-	-	-	-	-	-	-	-	-	-	
REFELURA	0.00=	0.010	0.005	0.090	1.000	•	~	•	•	~	<del>-</del>	-	-	~	-	-		~			- 4	-	-		
-DEFENDER:	0.00=	U.025	-0.015	0.065	0.010	1.000		-	-			-		~	-	-									
-PROSPECTOR	-0.014	0.040	0.058	40.054	-0.030	-0.0ye	1.000	-	-		<del>-</del>			~		-						-			
C-SIZE®	0.000	40.070	-0.09	0.650	V.100	0.032	-0.10=	1.000			۳										¥				
-ROGE	40.095	V.330	0.27/8	0.045	0.010	0.025	0.075	-0.290	1.000		٠														
URaile	40.004	U.190	0.308	0.050	70.025	40,055	0.088	40.250	0.162	1.008	۳										۳				
1CAMITALE	0.000	0.100	40.025	40.035	40.010	0.025	-0.025	-0.100	0.250	-0.084	1000														
27-DEVERAGES	0.00	-u.ore	40.056	0.204	0.034	70.050	0.025	0.180	-0.100	-0.22	0.085	1.000			۳										
E-PINIDANG#	0.098	40.035	-0.025	0.098	0.056	70.025	0.098	40.076	77.060	0.095	40.185	0.214	1.000												
4WVXEE	0.025	0.030	-0.002	-0.055	0.064	TV.130	40,056	-ar.use	U.120	40.1150	00728	0.196	-0.220	1.008											
5NOLE	-0.00	0.014	0.028	40.074	0.006	T0.030	-0.002	-0.19s	0.140	0.062	00038	-acore	0.095	0.014	1,000										
6YOROP												40.104													
7HEED PCT												0.030													
87-DACCE												0.036													
VCECCORDS																			1.00=						
051808D0=																			0.130						
IIWEDOOF			i i																						ļ
			i																0.374						ļ
77BAIZE0	i																		-U.37¢					-	
ISWOULDING			i																-0.035						
4ACSIZES	0.015	-0.052	-0.070	0.20=	0.045	0.06=	-0.030	0.332	-0.10=	-0.170	-0.012	0.078	-0.000	0.062	-0.11=	-0.010	-0.100	0.02=	-U.22°	0.09=	-0.130	0.37/2	0.16=	1.00=	

#### Notes to Table 2:

Pearson correlation coefficients appear in the lower diagonal. Bold font indicates a statistically significant correlation at the 5% level.

CASH\_ETR is cash taxes paid divided by pretax book income. Cash ETRs with negative denominators are deleted. The remaining non-missing ETRs are winsorized (reset) so that the largest observation is equal to 1 and the smallest is equal to 0. This variable is industry-adjusted by subtracting the industry mean (by year and two-digit SIC code). BTD is total book-tax differences, defined as pretax income less estimated taxable income, scaled by lagged assets. Estimated taxable income is the sum of current federal tax expense and current foreign tax expense divided by the top U.S. statutory tax rate. If current federal tax expense is missing, then we calculate it as total tax expense minus deferred tax expense, state income taxes, and current foreign tax expense. This variable is industry-adjusted by subtracting the industry mean (by year and two-digit SIC code). PERMDIFF is the difference between total book-tax differences (defined as in BTD) and temporary book-tax

differences, scaled by beginning of the year total assets. Temporary book-tax differences are deferred tax expense divided by the top U.S. statutory tax rate. This variable is industry-adjusted by subtracting the industry mean (by year and two-digit SIC code). UTB is the log value of one plus the increase in current uncertain tax positions, industry-adjusted by subtracting the industry mean (by year and two-digit SIC code). %FEDIR is the percentage of independent financial experts on the audit committee. DEFENDER is an indicator variable coded 1 if the STRATEGY score for firm i in year t ranges from 6 (the minimum) to 12, 0 otherwise. PROSPECTOR is an indicator variable coded 1 if the STRATEGY score for firm i in year t ranges from 24 to 30 (the maximum), 0 otherwise. STRATEGY is the firm strategy score for firm i in year t, measured following Bentley et al. (2013) and Higgins et al. (2015). The maximum (minimum) value for this variable is 30 (6). SIZE is the log value of lagged total assets. ROA is pre-tax income, scaled by lagged total assets. R&D is research and development expense, scaled by lagged total assets. When XRD is missing, we set the value to 0. CAPITAL is capital expenditures scaled by lagged total assets. LEVERAGE is the sum of long-term debt and long-term debt in current liabilities, scaled by lagged total assets. INTANG is intangible assets, scaled by lagged total assets. PPNE is gross property, plant and equipment, scaled by lagged total assets. NOL is an indicator variable coded 1 for firm-years with non-missing and non-zero values for tax loss carryforward, 0 otherwise. FOROP is an indicator variable coded 1 for firm-years with non-missing and non-zero values for pretax foreign income, 0 otherwise. HELD PCT is the percentage of outstanding shares held by institutions (from Thomson-Reuters 13f Holdings). DACC is adjusted discretionary accruals based on the performance-adjusted modified cross-sectional Jones model, measured following Frank et al. (2009) and Kothari et al. (2005). CEOCOMP is CEO total compensation, scaled by lagged total assets. STKMIX is CEO stock-based compensation as a fraction of total compensation. PRETAXCF is pretax cash flows, defined as cash flows from operating activities plus taxes paid, scaled by lagged total assets. BSIZE is the log value of the number of directors on the board. **%OUTDIR** is the percentage of independent directors on the board. **ACSIZE** is the log value of the number of audit committee members. 100%INDAUD is an indicator variable coded 1 if the audit committee is composed 100% of independent directors, 0 otherwise. See Appendix A for variable definitions.

									le3¤											
									m-Result.											
			Fi	nancial i	Expert	ise	on the Au	dit.	Committe	ee-and	·Bu	siness Str	ate	gju						
							Fu.		ample=											
•	•	ķ	٩	¥	¥	¥	ħ	'n	ħ	'n	Ħ	ħ	я	'n	¥	'n	ħ	'n	'n	¥
ħ	٩															ъ				
<sup>†</sup>	٩	CASI	I_E	TR□	٩	92	≒ B		0	Ħ	я	PERMDIFF:			¥	'n	U'	ΓB¤		Ħ
ħ	Ħ	9,	¥	ħ	Ħ	9	ħ	Ä		Ħ	Ħ	'n	¥	ħ	Ħ	ħ	ħ	¥	ħ	'n
Independent·Variables□	ъ	Coeff.c	В	t-state	Ħ	'n	Coeff.□	В	t-state	'n	Ħ	Coeff.□	В	t-state	¥	'n	Coeff.□	В	t-state	¥
Hypothesized-Variables=		9.	ħ	9	Ħ	'n	ħ	Ħ	%	Ħ	Ħ	'n	'n	°p	¥	ħ	ħ	Ħ	'n	Ħ
%FEDIR¤	Ħ	0.0052⊭	'n	0.66≒	Ħ	'n	-0.0028¤	Ħ	-1.22¤	Ħ	я	-0.0030¤	Ħ	-1.11b	¥	ħ	-0.0580¤	'n	-1.16¤	Ħ
%FEDIR*DEFENDER¤	٩	-0.0662	'n	-2.92¤	****	9	0.0070⊭	я	1.18¤	Ħ	91	0.0113≒	я	1.469	Ħ	9	0.3014	٩	2.23≒	**
%FEDIR*PROSPECTOR	Ħ	0.0292⊭	ħ	1.16	Ħ	ħ	-0.0171≒	я	-1.74¤	*¤	я	-0.0329¤	Ħ	-2.56¤	**¤	ħ	-0.4659¤	Ħ	-3.09¤	**
DEFENDER¤	Ħ	0.0354	'n	2.18¤	***	ħ	-0.0043¤	я	-1.05¤	Ħ	я	-0.0103¤	¥	-1.95¤	*¤	9	-0.1183¤	'n	-1.28¤	'n
PROSPECTOR¤	Ħ	-0.0207	'n	-1.26≒	¥	9	0.0079⊭	я	1.25¤	Ħ	94	0.0164¤	Ħ	2.02≒	**¤	9	0.1830¤	٩	1.95≒	•¤
Control-Variables=	п	٩	'n	9	¥	95	¥	95	ħ	Ħ	я	ħ	92	ъ	¥	ħ	9	٩	*	¥
SIZE¤	Ħ	-0.0074:	'n	-3.66¤	****	ъ	0.0005⊭	Ħ	0.81¤	Ħ	Ħ	0.0029¤	£	4.23¤	****	ħ	0.5753¤	Ħ	37.99	**
ROA¤	ħ	-0.8446	'n	-16.50:	***	я	0.6555≒	я	45.38¤	****	9;	0.5056¤	95	27.05	****	95	0.0744	٩	0.45⊭	¥
R&D¤	ħ	-0.3862	'n	-8.09⊭	***	95	0.0323⊭	Ħ	1.69¤	•¤	9	0.0601	ъ	2.51≒	**¤	ъ	4.1933¤	٩	17.74	**
CAPITAL	ħ	-0.0805	9	-1.63¤	¥	ъ	0.0232⊭	я	1.26	Ħ	9	-0.0503¤	95	-2.56¤	••¤	ъ	0.2829¤	91	0.78⊭	Ħ
LEVERAGE¤	ħ	-0.0270	'n	-2.87¤	***	ъ	0.0087⊭	я	2.25¤	***	9	0.0081¤	'n	1.83≒	*¤	я	-0.04084	٩	-0.60⊭	95
INTANG¤	ħ	0.0206	٩	2.24	•••	ъ	0.0046⊭	94	1.59¤	ħ	9	0.0030⊭	9	0.93⊨	¥	ъ	0.3864¤	9	6.11¤	**
PPNE¤	ħ	-0.0105	'n	-1.42¤	¥	ъ	0.0120⊭	94	5.29¤	***	9	0.0096⊭	9	3.60⊭	***	95	-0.0785¤	٩	-1.68¤	•9
NOL¤	ħ	-0.0052	9	-1.29s	ъ	ъ	0.0001¤	9	0.07⊭	ъ	9	0.0002≿	9	0.16	ъ	ъ	-0.01624	9	-0.60¤	9
FOROPa	ħ	0.0225	'n	5.04⊭	***	9	-0.0017:	£	-1.28¤	Ħ	9	0.0015¤	9	0.98¤	¥	'n	0.2295¤	9	8.27⊭	
HELD_PCT¤	'n	-0.0096	٩	-0.95⊭	¥	91	0.0003≒	٩	3.97⊭	****	9	0.0002⊭	я	3.11≒	***	'n	-0.30124	'n	-4.76¤	**
DACC	ħ	0.0013		1.07⊭	¥	я	0.0000¤	£	-0.08¤	ħ	9	-0.00024	뮢	-0.41¤	٩	9	-0.0080¤	Ħ	-1.18¤	Ħ
CEOCOMP¤	Ħ	-0.0038	٩	-4.98¤	***	я	-0.0015	£	-4.71¤	****	9	-0.0006=	я	-1.70¤	*¤	я	0.0157⊭	Ħ	3.70⊭	**
STKMIX¤	<u>-</u> -	-0.0028		-0.36⊭	¥	я		;	5.19¤	****	9	0.0048¤	핗	1.80¤	*g	9,	-0.1244	ъ	-2.25¤	**
PRETAXCF¤	'n	0.70846		16.31⊭	***	9	-0.3005⊭	4	-19.87¤	***	9	-0.1700⊨	핗	-9.47s		÷	2.0092⊭	9	10.33⊭	**
BSIZE¤	Ħ	-0.0070	9	-0.78⊭	%	4	0.0021⊭	4	0.76⊭	ħ	9	-0.0053¤	핗	-1.79¤	*g	9	0.2268¤	9	4.03⊭	**
%OUTDIR¤	ħ	0.0047		0.32¤		£	0.0032¤	ħ	0.70⊨	ħ	9	-0.0001¤	퓦	-0.02¤	٩	£	0.4005¤	¥	4.12⊠	**
ACSIZE¤	Ħ	0.0174⊭	٩	1.73≒	*¤	٩	-0.0128=	'n	-4.28¤	***	95	-0.0117⊭	я	-3.51¤	****	9	-0.1034a	<u>-</u>	-1.50¤	٩
100%INDAUD¤	ħ	0.0089	ъ	1.49¤	¥	ъ	0.0022≒	ъ	1.24¤	ħ	뭥	0.0025⊭	맺	1.19¤	٩	٩	0.0899¤	ħ	2.31□	**
<b>b</b>	9	ъ.	9	ъ	9	4	ħ	9	9	₩.	귛	ъ	핖	9	ħ	9	ħ	9	ħ	9
Να	95	···9.670⊭	9	9	9	9	9.108¤	4	9	9	륑.	···9.086⊭	핗	9	9	÷	6.090⊭	9	₩	9
F-stato	95	19.08 (			⊕	÷				₩ 1	-윙	58.64 (p			⊕	÷	187.53 (			
Adjusted R-squaredo		··0.055¢	•	8	Î		-0.544¤			- G		0.380⊭			- î		-0.494¤		8	9

## Notes to Table 3:

\*\*\*, \*\*, and \* indicate significance for a two-tailed test at the 1%, 5% and 10% levels, respectively. The reported standard errors are Huber-White robust standard errors.

The results are from estimation of the following model (subscripts suppressed):

TaxPlanning =  $\alpha + \theta_1\%$ FEDIR +  $\theta_2\%$ FEDIR\*DEFENDER +  $\theta_3\%$ FEDIR\*PROSPECTOR +  $\theta_4$ DEFENDER +  $\theta_5$ PROSPECTOR +  $\varphi$ Controls +  $\varepsilon$ 

TaxPlanning is CASH\_ETR, BTD, PERMDIFF, or UTB. CASH\_ETR is cash taxes paid divided by pretax book income. Cash ETRs with negative denominators are deleted. The remaining non-missing ETRs are winsorized (reset) so that the largest observation is equal to 1 and the smallest is equal to 0. This variable is industry-adjusted by subtracting the industry mean (by year and two-digit SIC code). BTD is total book-tax differences, defined as pretax income less estimated taxable income, scaled by

lagged assets. Estimated taxable income is the sum of current federal tax expense and current foreign tax expense divided by the top U.S. statutory tax rate. If current federal tax expense is missing, then we calculate it as total tax expense minus deferred tax expense, state income taxes, and current foreign tax expense. This variable is industry-adjusted by subtracting the industry mean (by year and two-digit SIC code). PERMDIFF is the difference between total book-tax differences (defined as in BTD) and temporary book-tax differences, scaled by beginning of the year total assets. Temporary book-tax differences are deferred tax expense divided by the top U.S. statutory tax rate. This variable is industry-adjusted by subtracting the industry mean (by year and two-digit SIC code). UTB is the log value of one plus the increase in current uncertain tax positions, industry-adjusted by subtracting the industry mean (by year and two-digit SIC code). %FEDIR is the percentage of independent financial experts on the audit committee. DEFENDER is an indicator variable coded 1 if the STRATEGY score for firm i in year t ranges from 6 (the minimum) to 12, 0 otherwise. PROSPECTOR is an indicator variable coded 1 if the STRATEGY score for firm i in year t ranges from 24 to 30 (the maximum), 0 otherwise. STRATEGY is the firm strategy score for firm i in year t, measured following Bentley et al. (2013) and Higgins et al. (2015). The maximum (minimum) value for this variable is 30 (6). Controls is a vector of control variables that includes the following: SIZE is the log value of lagged total assets. ROA is pre-tax income, scaled by lagged total assets. R&D is research and development expense, scaled by lagged total assets. When XRD is missing, we set the value to 0. CAPITAL is capital expenditures scaled by lagged total assets. LEVERAGE is the sum of long-term debt and long-term debt in current liabilities, scaled by lagged total assets. INTANG is intangible assets, scaled by lagged total assets. PPNE is gross property, plant and equipment, scaled by lagged total assets. NOL is an indicator variable coded 1 for firm-years with non-missing and nonzero values for tax loss carryforward, 0 otherwise. FOROP is an indicator variable coded 1 for firmyears with non-missing and non-zero values for pretax foreign income, 0 otherwise. HELD\_PCT is the percentage of outstanding shares held by institutions (from Thomson-Reuters 13f Holdings). DACC is adjusted discretionary accruals based on the performance-adjusted modified cross-sectional Jones model, measured following Frank et al. (2009) and Kothari et al. (2005). CEOCOMP is CEO total compensation, scaled by lagged total assets. STKMIX is CEO stock-based compensation as a fraction of total compensation. PRETAXCF is pretax cash flows, defined as cash flows from operating activities plus taxes paid, scaled by lagged total assets. BSIZE is the log value of the number of directors on the board. **%OUTDIR** is the percentage of independent directors on the board. **ACSIZE** is the log value of the number of audit committee members. 100%INDAUD is an indicator variable coded 1 if the audit committee is composed 100% of independent directors, 0 otherwise. See Appendix A for variable definitions.

								Γab	le∙4¤												
							Regres	sio	n-Resul	50											
			Fin	ancial l	Expert	ise (	on the Au	dit4	Commit	ee-an	d∙Bı	usiness St	rate	gyo							
						Def	enders an	d-P	rospect	ors-Oi	ηĮγο	1									
Ħ	91	9	91	94	¥	٩	<sup>9</sup>	٩	%	¥	9,	%	91	¥	¥	я	9	9;	%	¥	
Ħ	Ħ								Deper	dent-	Var	ariable¤									
₽	Ħ	CASE	ΙE	TR□	¥	¥	B3	ΓDο	1	%	Ħ	PERM		IFF≎	٩	94	UTB□			¥	
₽ .	Ħ	₹.	9	£	Ħ	9	ħ	ħ	'n	°ħ	95	ħ	٩	ħ	٩	94	¥	'n	¥	¥	
Independent Variables ::	ъ	Coeff.:	В	t-state	Ħ	ħ	Coeff.□	D	t-state	Ħ	я	Coeff.□	В	t-state	Ħ	9	Coeff.□	В	t-state	Ħ	
Hypothesized Variables=		9:	ъ	£	٩	9	ħ	Ħ	9±	٩	я	ħ	ъ	ħ	Ħ	я	Ħ	'n	9±	¥	
%FEDIR¤	ğ	-0.0652	٩	-2.84¤	***	92	0.0009¤	ħ	0.15¤	Ħ	ъ	0.0064¤	я	0.81¤	Ħ	94	0.1180¤	'n	0.93¤	Ħ	
%FEDIR*PROSPECTOR	Ħ	0.0908	٩	2.83≒	****	9	-0.0260¤	я	-2.51¤	**¤	я	-0.0445¤	Ħ	-3.26≒	***	¥	-0.56029	9	-2.93¤	**	
PROSPECTOR¤	ğ	-0.04979	Ħ	-2.15¤	**¤	ħ	0.0182¤	ъ	2.60⊭	***	я	0.0343¤	Ħ	3.87¤	***	9	0.1945¤	'n	1.51¤	¥	
Control-Variables=	•	9	Ħ	ъ	٩	9	9	ъ	7	ħ	я	95	Ħ	9	Ħ	¥	9	٩	9	Ħ	
SIZE¤	ğ	-0.0091	ħ	-1.84¤	*¤	ħ	0.0020⊨	ъ	1.45%	Ħ	я	0.0054¤	ħ	2.86¤	****	9	0.7263⊭	ħ	23.924	••	
ROA¤	ħ	-0.7907s	٩	-6.25¤	***	٩	0.7332¤	я	23.57	***	٩	0.6170⊭	ъ	17.30=	***	9	0.3785¤	'n	1.12⊨	¥	
R&D¤	ħ	-0.5442	95	-4.42¤	****	9	0.0332¤	ъ	0.81≒	Ħ	я	0.0819	'n	1.65	*¤	'n	2.5116	9	4.56¤	**	
CAPITAL:	ħ	0.0273⊭	٩	0.22⊭	¥	ħ	-0.04045	я	-1.03≒	Ħ	٩	-0.12824	'n	-2.49¤	***	٩	-1.37344	'n	-2.05¤	**	
LEVERAGE:	ħ	-0.0292	٩	-1.23¤	¥	ъ	-0.0010=	я	-0.11¤	Ħ	'n	-0.0109¤	ħ	-0.91¤	ħ	'n	0.0087¤	ъ	0.07⊭	¥	
INTANG¤	Ħ	0.0487	٩	2.17⊭	***	ъ	0.0046	9	0.56⊭	Ħ	92	-0.0041¤	ъ	-0.42¤	Ħ	ħ	0.1193¤	ъ	0.95≒	¥	
PPNE¤	Ħ	-0.0079	Ħ	-0.44¤	¥	ъ	0.0200≒	я	3.85⊭	****	'n	0.0210	Ħ	3.05⊭	***	9	-0.3289	ъ	-3.37¤	**	
NOL¤	'n	-0.01148	٩	-1.16¤	Ħ	ъ	0.0008¤	я	0.30⊭	Ħ	٩	-0.0048=	ъ	-1.37¤	Ħ	9	-0.1239¤	я	-2.20¤	**	
FOROP¤	'n	0.0225₺	٩	2.02⊭	***	я	-0.00584	я	-1.74¤	•¤	9,	-0.0016=	ъ	-0.39¤	Ħ	ħ	0.2199¤	я	3.54¤	**	
HELD_PCT¤	'n	-0.0412	٩	-1.62¤	¥	я	0.0294¤	я	3.60¤	***	٩	0.0269≒	Ħ	2.55¤	**¤	ħ	-0.5617¤	ч	-3.85≒	**	
DACC¤	ħ	0.0014	Ħ	0.66≒	٩	я	-0.0004	я	-0.54¤	ħ	٩	-0.0005¤	Ħ	-0.55¤	٩	ħ	0.0077¤	я	0.79¤		
CEOCOMP¤	ħ	-0.0051	Ħ	-2.73¤	***	9	-0.0010¤	я	-1.27¤	ħ	¥	-0.0008=	ъ	-0.74¤	Ħ	ħ	0.0526¤	ъ	5.25¤		
STKMIX¤	Ħ	-0.0158	Ħ	-0.79⊭	Ħ	Ħ	0.0001¤	Ħ	0.02⊠	ħ	ħ	-0.0015¤	Ħ	-0.20¤	Ħ	ъ	-0.34744	я	-2.72¤	**	
PRETAXCF¤	Ħ	0.6338	я	5.83¤	***	94	-0.2744	9	-7.87¤	****	ъ	-0.1587¤	'n	-4.48¤	***	8	1.2645¤	я	3.08¤	**	
BSIZE¤	٩	0.0010⊭	٩	0.04⊭	٩	я	0.0115¤	Ħ	1.749	•¤	٩	-0.0035	٩	-0.49¤	Ħ	я	-0.0426=	я	-0.35¤		
%OUTDIR¤	Ħ	0.0243⊭	Ħ	0.69¤	٩	я	0.0246¤	Ħ	2.35≒	**¤	ħ	0.0095⊨	'n	0.66≒	'n	я	0.4667⊭	я	2.27⊭	**	
ACSIZE¤	Ħ	0.0053⊭	Ħ	0.21≒	٩	я	-0.0138=	٩	-1.91¤	•¤	٩	-0.0140=	٩	-1.72¤	*¤	я	-0.03344	я	-0.23¤	Ħ	
100%INDAUD¤	ħ	0.0149	я	1.03⊭	٩	ъ	0.0007⊭	¥	0.18¤	ħ	ħ	-0.0006=	¥	-0.14¤	Ħ	8	0.1898¤	я	2.31¤	**	
ħ	'n	9	Ħ	9	ħ	9	ħ	ъ	%	'n	9	ħ	'n	₽.	Ħ	8	ħ	я	ħ	9	
Nα	95	-1,621¤	ъ	ħ	Ħ	٩	-1,610¤	ъ	Ħ	¥	9	-1,608¤	ħ	ħ	Ħ	я	-1,162¤	ъ	ħ	9	
F-state	95			).01)¤	٩	¥	43.69 (r		0.01)¤	ħ	ъ	23.19 (r	1	0.01)¤	ħ	я	76.04-0	0<(	).01)¤		
Adjusted R-squaredo	94	··0.068¤		9	ъ		-0.661¤		ъ	ъ		-0.520¤			ъ	94	0.597b		8	9	

## Notes to Table 4:

\*\*\*, \*\*, and \* indicate significance for a two-tailed test at the 1%, 5% and 10% levels, respectively. The reported standard errors are Huber-White robust standard errors.

The results are from estimation of the following model (subscripts suppressed):

TaxPlanning =  $\alpha + \theta_1\%$ FEDIR +  $\theta_2\%$ FEDIR\*PROSPECTOR +  $\theta_3$ PROSPECTOR +  $\gamma$ Controls +  $\varepsilon$ 

TaxPlanning is CASH\_ETR, BTD, PERMDIFF, or UTB. CASH\_ETR is cash taxes paid divided by pretax book income. Cash ETRs with negative denominators are deleted. The remaining non-missing ETRs are winsorized (reset) so that the largest observation is equal to 1 and the smallest is equal to 0. This variable is industry-adjusted by subtracting the industry mean (by year and two-digit SIC code). BTD is total book-tax differences, defined as pretax income less estimated taxable income, scaled by lagged assets. Estimated taxable income is the sum of current federal tax expense and current foreign tax expense divided by the top U.S. statutory tax rate. If current federal tax expense is missing, then we calculate it as total tax expense minus deferred tax expense, state income taxes,

and current foreign tax expense. This variable is industry-adjusted by subtracting the industry mean (by year and two-digit SIC code). PERMDIFF is the difference between total book-tax differences (defined as in BTD) and temporary book-tax differences, scaled by beginning of the year total assets. Temporary book-tax differences are deferred tax expense divided by the top U.S. statutory tax rate. This variable is industry-adjusted by subtracting the industry mean (by year and two-digit SIC code). UTB is the log value of one plus the increase in current uncertain tax positions, industry-adjusted by subtracting the industry mean (by year and two-digit SIC code). %FEDIR is the percentage of independent financial experts on the audit committee. PROSPECTOR is an indicator variable coded 1 if the STRATEGY score for firm i in year t ranges from 24 to 30 (the maximum), 0 otherwise. STRATEGY is the firm strategy score for firm i in year t, measured following Bentley et al. (2013) and Higgins et al. (2015). The maximum (minimum) value for this variable is 30 (6). Controls is a vector of control variables that includes the following: SIZE is the log value of lagged total assets. ROA is pretax income, scaled by lagged total assets. R&D is research and development expense, scaled by lagged total assets. When XRD is missing, we set the value to 0. CAPITAL is capital expenditures scaled by lagged total assets. LEVERAGE is the sum of long-term debt and long-term debt in current liabilities, scaled by lagged total assets. INTANG is intangible assets, scaled by lagged total assets. PPNE is gross property, plant and equipment, scaled by lagged total assets. NOL is an indicator variable coded 1 for firm-years with non-missing and non-zero values for tax loss carryforward, 0 otherwise. FOROP is an indicator variable coded 1 for firm-years with non-missing and non-zero values for pretax foreign income, 0 otherwise. HELD PCT is the percentage of outstanding shares held by institutions (from Thomson-Reuters 13f Holdings). DACC is adjusted discretionary accruals based on the performance-adjusted modified cross-sectional Jones model, measured following Frank et al. (2009) and Kothari et al. (2005). CEOCOMP is CEO total compensation, scaled by lagged total assets. STKMIX is CEO stock-based compensation as a fraction of total compensation. PRETAXCF is pretax cash flows, defined as cash flows from operating activities plus taxes paid, scaled by lagged total assets. BSIZE is the log value of the number of directors on the board. %OUTDIR is the percentage of independent directors on the board. ACSIZE is the log value of the number of audit committee members. 100%INDAUD is an indicator variable coded 1 if the audit committee is composed 100% of independent directors, 0 otherwise. See Appendix A for variable definitions.

							T	abl	e: 5¤											
							Regres	io	n-Results	Ħ										
Acc	our	nting- and-l	Nor	n-Accoun	ting- F	ino	ncial- Expe	rtis	e-on-the	-Audi	t-Co	mmittee · o	ind	Busines	s-Stra	teg	 γ⊭			
						Def	enders-and	j. p	rospecto	rs- On	y⊭									
и	Ħ	%	94	9	¥	91	¥	9	¥	¥	94	<sup>9</sup> t	91	¥	¥	¥	<sup>9</sup> #	¥	¥	¥
¥	95								Depe	ndent-	Va	riable¤								Ħ
<b>h</b>	9	CASH	L_E	TRX	¥	91	% BTD¤				% % PERMDIFF%				Ħ	¥	U	í	'n	
<sup>3</sup> t	91	°±	91	34	¥	91	¥	9	٩	'n	94	°¥	91	¥	'n	¥	<sup>9</sup> #	¥	¥	'n
Independent · Variables¤	31	<u>Coeff</u> , ¤	Ħ	t-stats	¥	'n	Coeff.¤	Ħ	t-stat#	¥	91	Coeff.¤	Ħ	t-state	¥	¥	<u>Coeff</u> ,¤	Ħ	t-statu	¥
Hypothesized∙ Variables⊭	Ħ	¥	91	95	¥	91	¥	9	°¤	¥	91	¥	91	¥	%	¥	¥	٩	¥	°¤
%ACCDIRM	98	-0.0681¤	91	-1.80¤	*¤	91	0.0018	9	0.18	¥	91	-0.0138¤	91	-1.10¤	¥	¥	-0.1373	¥	-0.66¤	¥
%NONACCDIR#	%	-0.068¤	91	-2.89¤	****	91	0.0022	9	0.35¤	¥	91	0.0137¤	91	1.72¤	*¤	¥	0.1784	¥	1.34¤	¥
%ACCDIR*PROSPECTORX	95	0.0797¤	91	1.42%	¥	91	-0.0184	9	-1.07¤	¥	91	-0.0175¤	91	-0.72¤	٩	¥	-0.1649	¥	-0.52¤	¥
%NONACCIR*PROSPECTORX	%	0.1018¤	91	2.91점	****	91	-0.0304	91	-2.80¤	***;	91	-0.0529¤	91	-3.94¤	****	¥	-0.7107	¥	-3.52¤	•••
PROSPECTORM	%	-0.0533¤	91	-2.30¤	**¤	91	0.0191	9	2.724	****	91	0.0344¤	91	3.88¤	****	¥	0.2009	¥	1.55¤	¥
<b>h</b>	%	'n	91	°#	¥	91	'n	91	94	9	91	'n	91	9	9	¥	3	¥	9	٩
Controls¤	٩	Yes¤	91	¥	¥	٩	-Yes¤	9	Ħ	¥	91	-Yes¤	91	¥	¥	¥	Yes¤	¥	9	٩
N¤	٩	1,621¤	91	¥	¥	91	-1,610¤	9	¥	¥	91	-1,608¤	91	¥	¥	¥	-1,162¤	¥	¥	•
F-stat¤	٩	5.33-(p	<-(	0.01)¤	¥	94	40.10-(	><	0.01)¤	¥	91	21.51-(	><	0.01)¤	'n	¥	70.92-(	p<-	0.01)¤	9
Adjusted·R-squared¤	٩±	0.067¤	¥	٩	٩	94	-0.661¤	9	٩	¥	93	0.521¤	٩	%	¥	'n	-0.599¤	¥	9	9

# Notes to Table 5:

\*\*\*, \*\*, and \* indicate significance for a two-tailed test at the 1%, 5% and 10% levels, respectively. The reported standard errors are Huber-White robust standard errors.

The results are from estimation of the following model (subscripts suppressed):

TaxPlanning =  $\alpha + \beta_1$ %ACCDIR +  $\beta_2$ %NONACCDIR +  $\beta_3$ %ACCDIR\*PROSPECTOR +  $\beta_4$ %NONACCDIR\*PROSPECTOR +  $\beta_5$ PROSPECTOR +  $\gamma$ Controls +  $\varepsilon$ 

TaxPlanning is CASH\_ETR, BTD, PERMDIFF, or UTB. CASH\_ETR is cash taxes paid divided by pretax book income. Cash ETRs with negative denominators are deleted. The remaining non-missing ETRs are winsorized (reset) so that the largest observation is equal to 1 and the smallest is equal to 0. This variable is industry-adjusted by subtracting the industry mean (by year and two-digit SIC code). BTD is total book-tax differences, defined as pretax income less estimated taxable income, scaled by lagged assets. Estimated taxable income is the sum of current federal tax expense and current foreign tax expense divided by the top U.S. statutory tax rate. If current federal tax expense is missing, then we calculate it as total tax expense minus deferred tax expense, state income taxes,

and current foreign tax expense. This variable is industry-adjusted by subtracting the industry mean (by year and two-digit SIC code). PERMDIFF is the difference between total book-tax differences (defined as in BTD) and temporary book-tax differences, scaled by beginning of the year total assets. Temporary book-tax differences are deferred tax expense divided by the top U.S. statutory tax rate. This variable is industry-adjusted by subtracting the industry mean (by year and two-digit SIC code). UTB is the log value of one plus the increase in current uncertain tax positions, industry-adjusted by subtracting the industry mean (by year and two-digit SIC code). %ACCDIR is the percentage of independent accounting experts on the audit committee. A director is defined as an accounting expert if he/she has work experience as a certified public accountant, auditor, CFO, financial comptroller, financial controller, or accounting officer. %NONACCDIR is the percentage of independent non-accounting financial experts on the audit committee. A director is defined as a non-accounting financial expert if he or she does not have work experience as a certified public accountant, auditor, CFO, financial comptroller, financial controller, or accounting officer but does have work experience as an investment banker, financial analyst, or any other financial management role, or experience supervising the preparation of financial statements (e.g., CEO or company president). PROSPECTOR is an indicator variable coded 1 if the STRATEGY score for firm i in year t ranges from 24 to 30 (the maximum), 0 otherwise. STRATEGY is the firm strategy score for firm i in year t, measured following Bentley et al. (2013) and Higgins et al. (2015). The maximum (minimum) value for this variable is 30 (6). Controls is a vector of control variables (untabulated). See Appendix A for variable definitions.